

Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 17340

Layoff Costs and Learning about Employer Financial Distress

Andreas R. Kostøl Matthew C. Merkle Morten H. Grindaker

OCTOBER 2024



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 17340

Layoff Costs and Learning about Employer Financial Distress

Andreas R. Kostøl BI Norwegian Business School, NTNU, Norges Bank and IZA

Matthew C. Merkle WP Carey School of Business, ASU

Morten H. Grindaker University of Chicago, Statistics Norway and IZA

OCTOBER 2024

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

ABSTRACT

Layoff Costs and Learning about Employer Financial Distress^{*}

While many employees risk losing their job and having their career disrupted due to employers' financial distress, it is widely recognized that many leave their employer in anticipation of layoff. In this paper, we assess how employee costs of financial distress depend on employees learning and acting on future layoff risk. To this end, we use random assignment of bankruptcy judges as an instrument for employer shutdown and administrative data on petition and guit dates to examine how earnings costs are shaped by employee reallocation. We show that shutdown causes a 24% fall in earnings over a fiveyear period despite one-quarter of employees having already left their firm. We document substantial heterogeneity in reallocation and earnings losses, typically displaying an inverse relationship, with higher reallocation in strong labor markets and from high-wage firms. The reallocation attenuates earnings losses by about 50%, approximately equal to the insurance from taxes and transfers. To assess the value of information, we estimate a model where risk averse workers learn about distress, search for jobs and access public insurance. Using the model, we calculate that employees' willingness to pay for their current job increases by 14% when the firm is liquidated without any advance information. Our findings suggest that making firms' financial risk information more accessible to employees can yield important benefits.

JEL Classification:	J62, J63, J65, J68, G33, G51, G52
Keywords:	job loss, search, occupational choice, bankruptcy,
	unemployment insurance

Corresponding author:

Andreas Ravndal Kostøl BI Norwegian Business School Nydalsveien 37 0484 Oslo Norway E-mail: andreas.r.kostol@bi.no

^{*} This paper should not be reported as representing the views of Norges Bank. The views expressed are those of the authors and do not necessarily reflect those of Norges Bank. This project received financial support from the Norwegian Research Council through grant 315437. We would like to thank David Autor, Richard Blundell, Carter Braxton, Sydnee Caldwell, Paula Calvo, Jonas Cederlof, Winnie van Dijk, Kyle Herkenhoff, Giulia Giupponi, Edwin Leuven, David Matsa, Magne Mogstad, Rafaelle Saggio, Michael Simmons, Isaac Sorkin, Chris Taber, Samuel Young and seminar participants at ASU, Boston College, Brandeis University, IFAU, The Norwegian Business School, NHH, NTNU, Society of Labor Economists, UCL-CREAM, University of Bergen, University of Copenhagen, University of Manchester, University of Oslo, and UW-Madision for useful comments and suggestions. We would also like to thank Hilde Ruus and Knut Ro for help with institutional details. All errors are our own.

1 Introduction

Each year, approximately two percent of the US workforce risk losing their job due to employer financial distress. Figure 1 illustrates the fraction of jobs at risk in bankruptcy, peaking at almost five percent during the Great Recession, and falling to one and a half percent in the aftermath of the pandemic. While bankruptcy in Chapter 7 and 11 usually means that a large fraction of employees are laid off, often leading to long-lasting earnings losses (see, e.g., Lachowska *et al.*, 2020; Bertheau *et al.*, 2022; Schmieder *et al.*, 2023), it is also widely recognized that many workers leave in anticipation of future layoffs (e.g., Stephens, 2004, and Baghai *et al.*, 2021). Understanding the extent to which labor markets insure and absorb risk matters for optimal policy and debates over disruption costs from future technological advancements (Agrawal *et al.*, 2023).



Figure 1: Employee Bankruptcy Risk in the US

However, assessing the employee costs of financial distress has proven difficult due to an adverse selection of employers and employees. First, employers in declining industries or firms that underperform relative to peers are more likely to go bankrupt. A consequence of this attrition is that workers in the least productive firms are more likely to experience the loss of an employer. Second, employers are likely to target the least profitable employees to lay off (e.g., Gibbons & Katz, 1991). Even when all workers remaining in the firm are laid off, as in Chapter 7 bankruptcy, learning that the job is at risk can induce quits in anticipation of costly layoffs. When workers with better outside options are more likely to leave, employees experiencing layoff may be those who are particularly disadvantaged.

In this paper, we address this selection using features of the Norwegian context that allow us to control for the quality of employers and employees who leave the firm before layoff begins. Our research design combines the random assignment of bankruptcy judges with administrative records of employment relationships, job openings and balance sheet information for employees and employers. There are three key ingredients to our empirical strategy. First, judges differ in their likelihood to liquidate otherwise similar firms, which allows us to address the selection of worker and employer quality. Second, information on creditor petition and employee quit dates allows us to account for workers who leave the firm before the petition is received. Third, we link

Notes: Own calculations using data from US Courts, CPS and Bernstein et al. (2019b). See Appendix B.3 for details.

this information with rich administrative data to perform a comprehensive analysis of the causal effects of shutting down a firm.

Our empirical analysis delivers three main conclusions. First, the loss of an employer causes a substantial loss to employees in firms that shut down. On average, liquidating a firm reduces earnings by one-quarter of pre-petition earnings over a five-year period. Second, job mobility rates in marginal, but non-liquidated firms, rise sharply the month the petition is filed – showing that costs are substantial even when many workers learn about the financial distress and switch employers before layoffs. Third, there is substantial heterogeneity in reallocation rates and size of earnings losses. While there are no differences in reallocation rates across skills or age groups, employees in high-wage firms and in stronger local labor markets leave their distressed employers at higher rates. Moreover, the size of earnings losses is inversely related to pre-bankruptcy employee reallocation rates: worker turnover rates are three times higher and earnings losses are halved when comparing outcomes after financial distress in a strong vs. weak labor market. In comparison, public insurance attenuates the net income loss by a roughly equal amount.

Understanding the welfare costs of financial distress requires a framework to account for how workers learn about employers' financial health and their labor market opportunities. Employees can only respond to layoff risks if they are aware of the financial circumstance of their employer. Moreover, their response depends on the value of a worker's current employer relative to alternative jobs: While workers with more to lose have a stronger motive to spend resources and effort to find new jobs to avoid future layoff, workers with less to lose from layoff may be more likely to find an equally well-paid job in the outside market (e.g., Caldwell & Danieli, 2023).

We address these conceptual challenges by developing and estimating a job ladder model. Central to our framework is that employees can learn about financial distress and face different local labor market conditions, in addition to making savings and job search decisions. Gross wages are determined by the heterogeneous productivity of an employer (i.e., a job ladder) and human capital—which accumulates over time—and income available for consumption is net after progressive taxes and unemployment insurance. We estimate the model using a combination of key population statistics and the causal evidence in two steps. First, some parameters are estimated directly from data. Second, we estimate the remaining parameters using the method of simulated moments to match our quasi-experimental findings. The model successfully replicates our key findings and matches several untargeted moments and estimates from empirical studies of unemployment benefits.

Next, we use the model to quantify the worker's willingness-to-pay (WTP) to keep the employer. Our WTP measure is the level of assets workers in liquidated firms would be willing to forgo to have their firm's bankruptcy case dismissed at the *time the judge makes the decision*. This captures not only the net present value of income lost from employer shutdown but also accounts for risk aversion and the cost of future search and expected human capital depreciation as workers foresee having to re-climb the job ladder. On average, the WTP is approximately equal to the net present value of the income loss but varies by the strength of the labor market. We find that the average WTP decreases by 0.21 percent for every one percent increase in the ratio of job openings to unemployed job seekers.

To explore the value of information and how it affects turnover dynamics, we conduct a counterfactual policy experiment that removes information about future layoffs. We label this experiment as "ideal" because it isolates the full value of an employer while also capturing the value of workers who might leave the firm before shutdown, irrespective of whether the employment relationships are of high- or low-value to the employee. This experiment allows us to calculate the insurance from information and labor market search by comparing the aggregate WTP with and without learning. Overall, we find that information reduces the welfare costs of layoffs by 12 percent. Next, we compare the wage losses of workers who stay with their employer until liquidation to the overall wage losses observed in the ideal experiment and the overall wage losses using our two-stage least-squares approach. This comparison allows us to calculate an implied bias from layoff regressions, arising from employees with relatively good outside options self-selecting out of distressed firms. On average, we calculate that this selection of high-value employment relationships increases the "measured earnings loss" by 10 percent.

Our paper advances research on sources of earnings losses following employer distress (see, e.g., Davis & von Wachter, 2011, and Carrington & Fallick, 2017 for reviews). In recent work, Lachowska *et al.* (2020) show that the loss of match-specific wage components are important in the US, while Schmieder *et al.* (2023) and Bertheau *et al.* (2022) find that employers play a more important role explaining earnings losses in Europe.¹ In a closely related paper, Rose & Shem-Tov (2023) first document earnings losses from mass layoffs in low-wage jobs, and then calibrate a dynamic job ladder model to quantify the willingness to pay to keep the job in the US. They find that workers are willing to pay more than half an annual salary to keep a job, indicating important frictions even in low-wage jobs.² Stepner (2019) shows that taxes play an important role mitigating the economic costs of layoff in the US, and Andersen *et al.* (2023) show that households cut down on private savings to maintain consumption levels after a mass layoff in Denmark.

Relative to existing research, we make several distinct contributions. First, we are the first to study how self-insurance from job mobility shape the earnings costs, and how it interacts with the

¹Our paper is thematically related to research on the consequences of corporate bankruptcy, aiming to understand how employee costs in bankruptcy shape a firm's financing decisions (see Matsa, 2018 for a survey). In two closely related papers, Araujo *et al.* (2020) and Bonfim & Nogueira (2021) show that reorganization leads to higher employee post-bankruptcy wages than when firms are liquidated in the Brazilian and Portuguese bankruptcy systems. We contribute to this research by showing that accounting for public and private insurance is crucial to quantify the employee costs of financial distress, consistent with Agrawal & Matsa (2013), who show expanding unemployment insurance coverage can allow for more leveraged firms. Caggese *et al.* (2019) and Baghai *et al.* (2020) study worker reallocation in firms facing financial distress in Sweden.

²Huckfeldt (2022) shows that excess earnings loss during recessions is primarily driven by workers who switch to occupations that pay lower wages—consistent with a model of occupation-specific and endogenous hiring standards. Jarosch (2022) shows that much of the earnings loss comes from repeated separations from new jobs in non-desirable firms. Our human capital process is similar to Ljungqvist & Sargent (1998), Herkenhoff *et al.* (2023), and Huckfeldt (2022), but differs by letting human capital be specific to a career (i.e., it is lost when interrupted by involuntary layoff), which weakens the incentives to invest in the labor market as people near retirement age. Our quantitative assessment relates to Burdett *et al.* (2020) and Audoly *et al.* (2022), who incorporate general and specific skills in equilibrium models with on-the-job search to rationalize the earnings losses upon layoffs. We extend their frameworks by including savings, as in Lise (2013), and by allowing for advance information about layoff risks.

tax and transfer system. Second, we offer a clear source of randomness to layoff experiences and an event date that allows us to account for pre-layoff attentuation.³ This combination allows us to address a lingering concern that employee turnover affects the composition of workers experiencing a layoff.⁴ Our paper is also related to a recent literature on expectations and informational policies in labor markets (see Mueller & Spinnewijn, 2023 for a review). Cederlöf *et al.* (2021) show that longer periods of mandatory notice increase the time spent searching for a new job. Other papers have shown that many workers learn about risks informally. Brown & Matsa (2016) show that financially distressed firms receive fewer job applications, and Baghai *et al.* (2020) show that high-skill workers are more likely to leave in anticipation of bankruptcy. Hendren (2017) shows that knowledge of future unemployment risk affects the pool of workers entering unemployment, which understates the value of UI. In contrast, we show that advance information affects the outmigration from firms in distress—overstating the value of public insurance.⁵

We contribute to a large body of research studying the degree to which individuals are insured against unemployment and income shocks (see Meghir & Pistaferri, 2011 for a review). One branch of this research focuses on the impact of income shocks or unemployment on consumption expenditure, with seminal contributions by Cutler & Katz (1992) and Blundell *et al.* (2008). We expand on existing research by showing that anticipatory turnover and private savings also play an important role in insuring against layoff costs.⁶ Another branch uses structural frameworks to examine the implications of unemployment insurance (UI), exemplified by the work of Saporta-Eksten (2014) and Braxton *et al.* (2020). While Low *et al.* (2010), Friedrich *et al.* (2019), and Liu (2019) extend the partial insurance framework with on-the-job search, the most closely related papers to our work are Lise (2013), who incorporates precautionary savings within a job ladder model, and Simmons (2021), who extends the job ladder model by allowing for partial information about layoff risk. We extend these papers by integrating a model of learning with both private and public insurance, and by providing evidence consistent with substantial insurance from job search.

³Our paper is related methodologically to several papers that use random assignment of judges and case examiners to examine a range of economic questions (see, e.g., Kling, 2006, Doyle, 2007, Aizer & Joseph J. Doyle (2013), Dahl *et al.*, 2014, Dobbie & Song, 2015, Galasso & Schankerman, 2015, Bhuller *et al.*, 2020, and Humphries *et al.*, 2019).

⁴Our results are consistent with the findings in Caldwell & Danieli (2023), who show that few outside options can explain a substantial fraction of earnings losses, and Cederlöf (2020), who shows that large layoffs lead to more pronounced earnings losses. Others have examined the implications of layoff misreporting for the empirical evidence (e.g., Flaaen *et al.*, 2019 and Birinci *et al.*, 2023).

⁵Ruhm (1992) studies the role of mandatory notification to employees targeted for layoff, and finds only small effects on non-employment spells. Ruhm (1994) finds that a combination of job counseling, training, and early warning leads to higher wages, while Cederlöf *et al.* (2021) provide evidence on a net productivity effect of mandatory notice in Sweden. Our evidence and framework is complementary to these studies. Ifergane *et al.* (2022) calibrates an equilibrium model of the labor market to quantify the insurance value of mandatory notice and understand how informational policies interact with traditional unemployment insurance.

⁶An extensive and growing body of research explores consumption changes around unemployment (see, e.g., Gruber, 1997, Chetty & Szeidl, 2007, and Ganong & Noel, 2019). Halla *et al.* (2020) and Andersen *et al.* (2023) find no role of the added worker effects following job displacement. Our paper also contributes to understanding the role of credit markets in mitigating the consequences of job loss. Keys (2018) shows that laid-off workers are more likely to go personally bankrupt, weakening the subsequent access to credit, while Braxton *et al.* (2020) show that existing credit lines provide valuable insurance to laid-off workers. Our evidence suggests that the credit market access mechanism remains available even after job loss but that people prefer to smooth consumption using assets rather than by increasing debt.

The remainder of the paper is organized as follows. Section 2 presents the data and empirical setting. Section 3 presents our instrumental variable strategy using judge fixed-effects to generate quasi-random variation in *employer shutdown*. Section 4 presents our main empirical findings. Section 5 develops a job search and consumption model and Section 6 brings it to the data. Section 7 uses the model to quantify the welfare implications of our quasi-experimental findings and to assess how worker turnover affects costs of financial distress. Section 8 concludes.

2 Institutional Setting, Data, and Descriptive Evidence

This section presents the Norwegian labor market institutions and the bankruptcy system, and describes the administrative data we will use.

2.1 Employment Protection and Unemployment Insurance

Employment protection in Norway is relatively strong, ranking about the median among OECD countries.⁷ Firms can hire employees on either fixed-term or permanent contracts. Resigning employees must notify the employer three months prior to leaving. Similarly, firms planning to downsize are required to notify workers, where dismissal time varies from one to six months depending on age and tenure. Wrongful discharge can result in a lawsuit where firms must compensate dismissed employees for lost income.

The unemployment insurance system is funded by payroll taxes and there is no experience rating on firms. The cash benefits compensate 62 percent of lost wage income but only up to six times the social security base amounts (Folketrygdens grunnbeløp, about NOK 100,000 in 2021, NOK/USD \approx 10). This feature creates a kink in the benefit schedule, similar to the US (see, e.g., Landais, 2015), where cash benefits are constant, and replacement rates decline non-linearly after this point.⁸ Appendix Figure A.1 plots unemployment rates in Norway from 2001 to 2016.

Layoffs due to bankruptcy differ from mass layoffs in a few important ways. First, there is no formal requirement of a notice period or dismissal time. Second, in contrast to mergers and acquisitions, workers lose tenure-based employment protection in bankruptcy. Third, a government wage guarantee scheme covers one month of unpaid wage bills and workers are eligible for unemployment insurance (UI) benefits immediately after bankruptcy is opened by the judge.⁹

⁷European labor laws regulate employment protection and mass layoffs. Union membership in Norway is relatively high compared to other countries in the OECD and the US. Still, it has fallen from 58 to 53 percent from 1992 to 2013 (OECD Statistics Trade Union Statistics, Accessed August 31st, 2020).

⁸The potential benefit duration is two years for everyone with previous income above two times the base amount. After UI benefits expire, workers may qualify for other transfers, such as vocational training or health-related benefits, typically replacing approximately two-thirds of lost income. The social assistance program is typically the last resort and is means-tested, with replacement rates varying across municipalities (see, e.g., Autor *et al.*, 2019).

⁹The wage insurance covers up to two base amounts takes (about NOK200,000) of non-paid wages the last six months and covers one month salary after the bankruptcy process starts. Retirement pensions are not held by the firm, but are managed by third-party financial institutions.

2.2 The Bankruptcy System

When a firm is temporarily or permanently unable to pay what it owes, it negotiates with its bank and other creditors outside the court system. The threat of liquidation is usually the creditors' outside option in the negotiation process. The debtor or its creditors (e.g., banks, suppliers, tax authorities, or employees) can file for bankruptcy, which is usually initiated after the cessation of payments and three months of unsuccessful debt enforcement. The tax authority and private creditors account for the vast majority of petitions.

Evaluating whether the firm is solvent requires discretion when assessing whether the financial distress is temporary and whether a firm's assets can be sold without business disruption. While these conditions apply uniformly to all firms, this discretion underlies significant variation in liquidation rates across judges, which we will use as part of our identification strategy in the next section. When the judge decides to open bankruptcy, a trustee is assigned to take control over the firm's assets and propose a plan to the judge. When the plan is approved, the trustee auctions the assets piece by piece or sells the firm to new owners, similar to Chapter 7 in the US system.¹⁰

Timing. Creditors must present documentation of whether the debtor firm is illiquid and insolvent. This information must be presented to the court in writing. When the petition is received in the district court, the case is assigned to a bankruptcy judge, who spends, on average, slightly more than one month examining the criteria for liquidating the firm. The case can be dismissed, meaning the firm can continue its operations, or the judge can start the liquidation process. Ninety-five percent of dismissed cases are processed within three months.

2.3 Data

Our empirical analysis is made possible by linking several data sources using anonymized identifiers of firms and their employees. The primary data source is the administrative employeremployee data, which offers detailed information about earnings and work contracts and covers transitions between firms from 1999 to 2018. We link this information with three additional data sources.

First—and central to our contribution—are the bankruptcy court files. These files cover every bankruptcy case filed from 2005 to 2018 and provide time stamps when cases enter the system. The data includes firm identifiers, anonymized identifiers for the assigned judge, and indicates whether or not the judge initiated the liquidation process. Second, we obtain information on individual wealth, and income sources—including bank deposits, capital income, and government

¹⁰Chapter 7 means liquidation through auction, with the goal of maximizing the recovery rate for creditors. The Norwegian bankruptcy system shares other similarities with the US system, for example, that cases are randomly assigned to judges (e.g., Bernstein *et al.*, 2019a). The priority of claims is the same, where administrative costs are paid before other claimants to the firm. Unpaid taxes, debts to government agencies, and wage claims have priority over secured creditors, and unsecured creditors get priority over shareholders. The major difference between the US and Norwegian bankruptcy systems is the use of reorganization. While this option is rarely used in Norway, Chapter 11 bankruptcy is the most common option among larger firms in the US. In 2019, 14,524 US firms filed for Chapter 7, while 5,814 filed for Chapter 11. Liquidation under US Chapter 7 is typically used by smaller firms (see, e.g., Table I in Bris *et al.*, 2006).

transfers—from tax records. We deflate income and wealth measures available from 1999 to 2017 to 2015-values using the consumer price index (CPI). To reduce noise from outliers, we winsorize market income and bank account statements at the 99th percentile and total debt and disposable income net of taxes at the 95th percentile. Third, we collect demographic information, such as gender, age, family composition, education of every resident since 1999, and participation in the social insurance program, such as unemployment insurance. The linked employer-employee data and income components from tax registers are third-party reported (e.g., by employers and financial intermediaries) and have been rated as exceptional by international data quality assessments (see, e.g. Atkinson *et al.*, 1995). Online Appendix B provides further details on the data sources. Lastly, we use data on vacancies from the Norwegian Welfare Administration that include information about the occupation and workplace location of the job opening.

2.4 Sample Construction and Summary Statistics

We impose four restrictions on the population of firms and employees. First, we use bankruptcy petitions from 2006 to 2013 to ensure a balanced sample over four years and a consistent measure of assets and income for the full sample.¹¹ Second, we restrict to full-time employees aged 25 to 62 to mitigate concerns about entry and exit from the labor force.¹² Third, we require that firms have at least five employees during the calendar year before the petition to focus on viable employers. Our last restriction is that workers must be employed at the firm 12 months before the petition. This choice provides us with the population at risk of layoff, allowing us to account for workers who leave the firm, voluntarily or involuntarily, before the petition date, and assess the role of anticipation effects and worker mobility.¹³

We impose two restrictions on the sample of district court judges. First, we require that a judge has handled at least 20 cases over the entire sample period. Second, we exclude court-year observations with fewer than five judges. These restrictions increase precision in the judge stringency measure and ensure a sufficient number of potential judges handling the case. Appendix B.2 describes the sample construction in detail. All control variables are measured the year before the petition.

Table 1 reports summary statistics for our estimation sample. We also compare our sample to the overall labor force and workers in the same age range starting an unemployment spell in 2011. Panel (a) reports averages and standard deviations for the workforce, and Panel (b) reports the

¹¹Due to changes in the measurement of real estate from a hedonic pricing model introduced from 2010 onward, we only have a consistent measure of gross wealth for eight years.

¹²The full-time employee restriction is common in the literature on job loss (see, e.g., Davis & von Wachter, 2011 and Lachowska *et al.*, 2020). Our data defines full-time employment as working more than 30 hours per week. We measure this at the end of the year before the petition. Existing research on layoff costs using administrative data typically requires the sample of employees to be long-tenured and the non-laid-off to be continuously employed by the distressed firm. We do not impose these restrictions.

¹³Two additional restrictions help reduce noise in our data. First, we exclude non-residents, which account for less than half a percent of the sample. This choice allows us to measure demographic variables. Second, we exclude manager-owners who are registered as employees. Managing-owners typically have different incentives and compensation schemes than regular employees.

same statistics for the unemployed. Panel (c) illustrates that our estimation sample is younger and has slightly lower income and liquid assets in bank accounts than the labor force but is older and has higher wage and disposable income than the unemployed. We see that unemployment and distress risk is higher for people with only a high school diploma or less. In contrast, Appendix Table C.1 reports summary statistics for long-tenured employees experiencing mass layoffs. The displacement sample is older and significantly higher paid than our sample and the overall workforce. The average firm in our sample has 22 employees. Because the firm size distribution has a long right tail, the average employee works for a firm with 153 employees. We illustrate the firm size distribution in Appendix Figure A.2.

Sample:	(a) Labo	or Force	(b) Une	mployed	(c) Emp	loyees in	
ownip to	Aged			25-62	Distressed Firms		
	Mean	St. Dev	Mean	St. Dev	Mean	St. Dev	
	(1)	(2)	(3)	(4)	(5)	(6)	
Age	42.4	10.9	37.7	11.2	40.3	10.0	
Female	0.409		0.347		0.233		
High School Graduate	0.356		0.376		0.433		
Bachelor Degree	0.283		0.157		0.152		
Higher Degree	0.110		0.055		0.042		
Wage Income (NOK1,000)	477.502	300.689	357.829	153.951	461.557	277.399	
Disposable Income (NOK1,000)	366.016	303.528	285.013	128.000	361.436	211.676	
Liquid Assets (NOK1,000)	146.978	286.620	86.668	198.372	101.637	202.894	
Industry Shares							
Mining, utilities, construction	0.137		0.179		0.199		
Manufacturing	0.120		0.212		0.222		
Trade and transportation	0.188		0.190		0.255		
Information, finance, prof. services	0.167		0.126		0.218		
Educational and health services	0.243		0.129		0.018		
Arts, recreation, hospitality services	0.030		0.060		0.055		
Number of observations	1,484,242		26,821		14,830		

Table 1: Summary Statistics

Notes: This table reports summary statistics of our estimation samples. Panel (a) reports averages and standard deviations of key characteristics, income, and wealth of the labor force in 2011. Panel (b) reports the same characteristics for unemployed individuals whose spell started in 2011. Panel (c) reports statistics for full-time employees aged 25–62 in firms petitioned for bankruptcy in 2006–2013. Nominal values are deflated to 2015-values using the consumer price index. NOK/USD ≈ 10 .

3 Measuring the Effects of Employer Distress

This section describes how we use judge assignment to address the challenges of identifying the labor market effects of employer shutdown.

3.1 Identification Challenge

Consider the treatment indicator, L_i , equal to 1 when employee *i*'s employer is liquidated and otherwise equal to 0. The outcome of interest y_i can be denoted as the potential outcome y_{0i} , observed if the employee is untreated, and by y_{1i} , observed when the individual loses her job. The target parameter is the individual treatment effect—the costs of employer shutdown—defined by $\beta_i = y_{1i} - y_{0i}$.

The identification challenge arises due to observing only one of these outcomes for each individual. Suppose, for example, that employees with low productivity (i.e., both y_{1i} and y_{0i} are low) are more likely to be employed by firms that go bankrupt. A similar selection of workers arise if workers with good outside options tend to have high values of both potential outcomes. This type of selection would lead to large differences between treated and non-treated workers, confounding selection with treatment effects. We address this challenge by using an instrumental variable that replaces the endogenous employment choices with plausibly exogenous variation in layoffs, and using the dates to include workers who leave in anticipation of bankruptcy.

3.2 Random Assignment of Bankruptcy Cases

Next, we describe the institutions that generate the plausibly exogenous variation in employer liquidation. Within each district court, the assignment of a case to judges is done by the chief judge.¹⁴ To ensure equal treatment by the law, the court rules state that case assignment shall be done "by drawing lots." In practice, the assignment is made without knowledge of case content on a rotating basis by the date the petition is received: Each time a new case arrives, it is mechanically assigned to the next judge on the list (see, e.g., Bohn, 2000 and Bhuller *et al.*, 2020).

Although we do not observe a judge's stringency directly, we can consistently estimate it using the bankruptcy petitions. Our measure of judge stringency is calculated for each case as the average rate at which the liquidation process is initiated in the other cases a judge has handled (i.e., a leave-one-out mean). Importantly, we use all cases—not only the estimation sample from 2006 to 2013—including firms without employees. Because the randomization occurs among judges within a district court, we use the residual variation in stringency after removing average liquidation rates within a court and given year.

To assess whether courts comply with the assignment protocol, we examine how judge stringency relates to the attributes of the firm and its employees. Columns 2 and 3 of Table 2 document that some worker and firm characteristics predict whether the firm is liquidated, and the joint test rejects the null hypothesis that they are not related to the outcome of the case. We also see that older or foreign-born workers are more likely to experience that the judge opens bankruptcy. Perhaps not surprisingly, we find that a one-year lag in total debt is a strong predictor of bankruptcy, but also note that debt is only available for firms with relatively high levels of revenue and at least 50 employees.

¹⁴The Norwegian court system is organized into three levels: the district, the appeals, and the supreme court. The vast majority of bankruptcy cases are settled in the district courts.

	(1)	(2)	(3)	(4)	(5)		
Dependent Variable:		Liquidatio	on of Firm	Judge Str	ingency (%)		
-	Average	coef.	s.e.	coef.	s.e.		
Panel (a):		Pre-determin	ed Worker C	Characteristi	cs		
Age (at the time of decision)	40.289	0.0040***	(0.0011)	0.0125	(0.0147)		
Female	0.233	-0.0133	(0.0378)	-0.2459	(0.4127)		
Married	0.469	0.0449**	(0.0195)	0.0373	(0.2649)		
Foreign born	0.225	-0.0309	(0.0392)	0.4051	(0.5304)		
Less than high school	0.373	-0.0054	(0.0263)	0.4514	(0.3317)		
High school	0.433	0.0149	(0.0227)	-0.2011	(0.3773)		
Bachelor degree	0.152	-0.0280	(0.0368)	-0.3938	(0.3701)		
Master degree	0.042	0.0300	(0.0531)	-0.1379	(0.6379)		
Tenure (months)	57.342	0.0004	(0.0003)	0.0029	(0.0029)		
Wage income (NOK1,000)	451,770	-0.0000	(0.0000)	-0.0007	(0.0011)		
Disposable income (NOK1,000)	345,892	-0.0000	(0.0000)	-0.0015	(0.0022)		
Liquid assets (NOK1,000)	101.638	0.0000	(0.0000)	-0.0007	(0.0005)		
Debt (NOK1,000)	950.327	-0.0000	(0.0000)	-0.0001	(0.0001)		
Panel (b):	Pre-determined Firm Outcomes						
Employment	153.296	-0.0001	(0.0003)	-0.0018	(0.0017)		
Revenue (NOK mill.)	363.114	0.0000	(0.0000)	-0.0000	(0.0002)		
Wage Bill (NOK mill.)	66.022	0.0000	(0.0005)	-0.0004	(0.0014)		
Total Debt (NOK mill.)	227.440	0.0001***	(0.0000)	-0.0001	(0.0002)		
Observations	14,826	14,826		14,826			
F-stat and p-value		33.9554	< 0.0001	1.1579	0.3012		

Table 2: Summary Statistics and Test of Random Assignment Mechanism

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Notes: This table reports summary statistics of the baseline estimation sample and tests of the random allocation procedure. There are 312 judges. All characteristics are measured one year prior to when the case was received in the court. Column 1 reports averages of pre-determined individual characteristics (Panel (a)) and pre-determined firm-variables (Panel (b)). OLS regression coefficients of these characteristics on a dummy variable for whether the firm was liquidated are reported in column 2 and on the measure of judge stringency in column 4. F-statistics are obtained from regressions on combined set of characteristics. The baseline sample of bankruptcy cases consists of petitions filed during 2006–2013 that are handled by judges with at least 20 cases. The sample includes firms with at least five employees before the petition is received, and full-time employees aged 25–62. Each regression controls for fully interacted year and court indicators, and a linear and quadratic term of the (leave-one-out) judge processing time. Revenue, total debt, and wage bill is in millions. Nominal values are deflated to 2015-values using the consumer price index. NOK/USD \approx 10.

Lastly, Columns 4 and 5 show no statistically significant relationship between any of the worker or firm characteristics and the stringency of the judge. Consistent with the random assignment mechanism stipulated by the law, we cannot reject that all characteristics jointly have no relationship with the stringency of the judge, with a p-value of 0.3.

3.3 Instrumental Variables Approach

We now describe how we use the judge fixed effect as an instrument for layoff. The key to our instrumental variable approach is that judges are both randomly assigned and that some judges are more likely to liquidate a firm than others. The research design uses this naturally occurring variation in liquidation propensities as a separation shock to otherwise stable employment relationships. We use panel data on wealth, income, and employment to study household consequences and implement the following instrumental variables (IV) model

$$L_i = \phi Z_{j(i)} + X'_i \psi + e_i \tag{1}$$

$$y_{it} = \beta_t L_i + X_i' \theta_t + u_{it}.$$
(2)

The instrument, $Z_{j(i)}$, is the stringency measure for judge j to which person i's employer is assigned. The vector X_i contains pre-determined observable characteristics for the person and always includes a constant and a full set of year-by-district-court dummies.¹⁵ The first stage coefficient ϕ gives the fraction of workers who lose their jobs when moving from a hypothetical judge with zero liquidation rate to a hypothetical judge that liquidates all cases. In the second stage equation, the outcome variable y_{it} is measured t years after the liquidation process starts (e.g., wage income four years after the decision). Due to the random assignment procedure, judge stringency is uncorrelated with equation 2's residual.

Our goal is to estimate the average of β_t among those separating from their employer because of the judge. We estimate the causal effect of layoff using the two-stage least squares (2SLS) estimator with first and second stage equations given by (1) and (2). The endogenous variable in this two-equation system indicates whether a worker is laid off because the firm is liquidated at time t = 0, rather than continuing with the employer or possibly separating at a later date for other reasons. Hence, the 2SLS estimate of β_t captures the effect of the *initial* layoff, which can operate through several channels, including loss of human capital, loss of tenure and the job protection rights associated with it, or shorter time to search out a good match in a new firm.¹⁶ Throughout our analysis, we will also decompose our treatment effect estimate into potential outcomes of treated and untreated workers in marginal firms (i.e., employees in non-liquidated firms). Imbens & Rubin (1997) develop this decomposition for discrete instruments with heterogeneous treatment effects and Dahl *et al.* (2014) extends the decomposition framework to settings with continuous instruments.

¹⁵To reduce residual noise in our outcome variables, we also include dummies for age groups, marital status, foreign born, education levels, and income quintiles.

¹⁶The effect could also be attenuated if employees of dismissed firms lose their job during subsequent restructuring or new bankruptcy petitions. We use the first case ever observed for firms in the court system and do not use any subsequent case(s). These cases are part of the outcome of continuing firms. We have explored the possibility of using this second experiment, but the sample is too small to yield precise inference. We do not observe informal negotiations with creditors or restructuring in the data.

Figure 2: Firm is Liquidated and Worker Separates



Notes: Panel 3a displays the relationship between judge stringency and firm liquidation (i.e., the first stage). Panel 3b displays the relationship between judge stringency and whether an employee separates from the distressed firm by the end of the petition year. The solid line plots a local linear regression of employment on judge stringency with a bandwidth of 0.2. Local regressions are conditional on fully interacted year and district court dummies. The histogram of judge stringency is shown in the background of both figures (top and bottom 1 percent excluded from the graph). Dashed lines are 95 percent confidence intervals.

Assessing the Instrument. We begin by assessing the relevance of the instrument. Figure 3a provides a graphical representation of the first stage of the IV model, where the background is a histogram of the empirical judge stringency distribution. The mean and median of the raw stringency variable is 0.63, with a standard deviation of 0.13. While controlling for fully interacted year and district-court dummies shrinks the standard deviation to 0.085, the range of judge stringency is still wide—with a 1st to 99th percentile range from 0.44 to 0.79.

We present the relationship between average liquidation probabilities and the firm's assigned judge by the solid black line and 95 percent confidence intervals with dashed lines. The lines are fitted from local linear regressions with a bandwidth of 0.2, showing that the relationship is monotonically increasing. Judges above the 90th percentile (i.e., 71 percent liquidation rate) liquidate about two-thirds of bankruptcy cases in our sample. In contrast, judges below the 10th percentile (i.e., 54 percent rate) liquidate about 45 percent of the distressed firms. Figure 3b plots the reduced form relationship between judge stringency and the likelihood an employee separates from their firm. While the relationship is somewhat flatter than liquidation, it shows that the judge stringency rate also raises separation rates—indicating a causal relationship between liquidation and job displacement.

Next, we turn to the regression counterpart of this figure. We estimate the first stage coefficient by regressing liquidation on judge stringency, controlling for fully interacted court and petition year fixed effects. We also partial out the covariance with processing time to be consistent with our empirical implementation in the next section. In Panel (a) of Table 3, we report the estimated first stage coefficient at the firm level. While each judge has handled an average of 92 cases, there are about six firms per judge in our empirical sample. This highlights that our judge fixed effect is estimated on a much larger sample than our estimation sample. We cluster standard errors at the

judge and petition year level and find the first stage coefficient is positive and statistically significant. A 10 percentage point increase in judge stringency leads to a 75 percentage point increase in the likelihood that a firm is liquidated. The second column restricts the sample to courts with at least five judges handling cases in a year. The point estimate is quantitatively similar but is more precisely estimated. In Panel (b), we repeat the exercise at the employee level. Moving to the third column, we show that the judge processing time does not significantly alter the first stage coefficient. The last column includes dummies for each age, marital status, education level, and immigrant status, and the third column includes a measure of judge experience. The two specifications show that the point estimate remains similar and statistically significant if we condition on a broad range of characteristics, demonstrating that the stringency measure captures important differences in how judges view the solvency question.

	First stage: Firm is Liquidated							
	(a) Firm	n Weights						
Sample / specification:	I. All	II. Baseline	I. All	II. Baseline	III. Processing	IV. Individual		
	Courts		Courts		Time	controls		
Judge stringency	0.753***	0.765***	0.818***	0.831***	0.781***	0.794***		
	(0.131)	(0.122)	(0.182)	(0.171)	(0.204)	(0.202)		
Mean liquidation	0.560	0.551	0.553	0.553	0.553	0.553		
Minimum Judges in Court	2	5	2	5	5	5		
Number of Judges:	357	312	357	312	312	312		
Avg. Cases per Judge:	87.8	92.2	87.8	92.2	92.2	92.2		
Number of Firms / Workers:	2,084	1,869	16,998	14,826	14,826	14,826		

Table 3: First Stage Estimates

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the first stage coefficients of equation 1. Baseline sample consists of petitions filed during 2006–2013 and handled by judges with at least 20 cases. The sample includes firms with at least five employees the year before the petition. The estimation sample includes full-time employees aged 25–62 employed 12 months prior to petition. There are 312 judges in the baseline sample. Regressions always controls for fully interacted year and court dummies. Individual controls include age, education, marital status, immigration indicators, and dummies for quintiles of pre-petition wage income levels. Judge average processing time is measured in all bankruptcy cases in the raw data.

To help interpret the instrument, we calculate the implied share of employees in marginal firms. This fraction can be obtained non-parametrically from Figure 2 as the difference in liquidation rates between the 99th and 1st percentile of judge stringency. We find the population share to be about 25 percent. In comparison, a relatively large share of firms (45 percent) always goes bankrupt irrespective of the judges they are assigned—seen in the figure as the liquidation rate at the lowest value of the stringency measure. The remaining fraction (i.e., 30 percent) are never-bankrupt firms whose outcome is independent of the judge. These firms are observed in court due to, for instance, overly aggressive creditors or automatic bankruptcy filings from tax authorities for missing annual reports.¹⁷ Under the linear specification, the fraction of marginal employees equals the first stage coefficient multiplied by the range of the stringency measure from Figure 3a. This specification

¹⁷We bootstrap the standard errors on the population share of compliers, always- and never-takers, and find the estimates very precise. The results are available upon request.

gives a share of marginal jobs of about 0.3. Appendix Table A.1 shows that the compliers are equally distributed across worker characteristics but are three times more likely to be employed by a firm above the median size (30 employees in the year prior to bankruptcy).

Additional Assumptions Interpreting the 2SLS estimates of equation (2) as the causal effects of *liquidation* requires two additional assumptions. The first assumption is that judge stringency affects outcomes of interest only through its impact on closing down the firm. This exclusion restriction appears likely in our setting. Since the bankruptcy outcome is limited to liquidating the firm or not, and there is no personal contact between the judges and the worker, the scope for the judge to affect employee outcomes directly is limited.¹⁸

However, judges may vary in how long they spend handling a case. In our setting, the correlation between the average liquidation rate and the processing time of judges is 0.38, raising a concern that judges who are more likely to liquidate would reinforce the layoff effects if processing time hurts workers' outcomes. Alternatively, time could be beneficial for workers as they have more opportunity to search for a better job. These potentially opposing forces means that ordering the two treatments is difficult, which may confound the causal interpretation of 2SLS estimates (see Mogstad *et al.*, 2021). A key feature in our data, however, is that processing time depends on the case's outcome: Only when the judge decides to open bankruptcy (i.e., liquidation) does the process significantly exceed one or two months. This feature allows us to examine how processing time affects outcomes by conditioning on liquidation.¹⁹ Our regression evidence indicates that the leave-out-average processing time does not have a significant impact on wages. However, we include average processing time and its square as controls in our regression analysis to alleviate any concerns of confounding effects.

The second assumption required for a causal interpretation of the IV estimates is that (a) treatment effects only vary by observable characteristics (i.e., constant treatment effects) or that (b) the relationship between layoff and the instrument is monotone. This monotonicity assumption requires that, for each firm, the probability of bankruptcy is at least as high if assigned to a strict judge (high value of $Z_{j(i)}$) as if assigned to a lenient judge (low value of $Z_{j(i)}$).²⁰ The monotonic

¹⁸The trustee would inform the court if there are reasons to impose a bankruptcy quarantine on board members or the CEO. Such a quarantine may be imposed if there is a suspicion of a criminal act in connection with the bankruptcy or the person is deemed unfit to establish a new company due to improper business conduct. Suppose the court decides to place, for example, the CEO in bankruptcy quarantine. In that case, s/he cannot serve as a board member or CEO for two years. As we have removed manager-owners from the sample, this judge outcome does not concern regular employees' outcomes.

¹⁹While 95 percent of dismissed cases are closed within 82 days, the mean processing time for firms being liquidated is 656 days, and the median is 472 days. We regress wages one year after on a set of dummies for court and petition year, age, income quintiles, gender, marital and immigration status, and a polynomial of judge processing time. Experimenting with linear, quadratic, and cubic specifications, we find the quadratic specification has a lower Akaike information criterion (AIC). This specification minimizes AIC when conditioning on liquidation and in the overall sample. With this specification, we find that 100 days of processing time reduces wages one year after the petition by 34,557, or about 7.5 percent of the average wage from Table 1. The point estimate of processing time is 346, with a clustered standard error of 278.2. The coefficient estimate on the processing time squared is -0.441, with a standard error of 0.355.

²⁰When the monotonicity assumption does not hold, individual weights (on β_i) may be negative, and the weights would not sum to 1. This would bias the results and, for example, yield a positive local average treatment effect (LATE)

relationship between the judge fixed effect and the outcomes in Figure 2 provides some support to the assumption. Another implication of monotonicity is that, in the absence of sampling error, first stage estimates should be non-negative for all subsamples. Again, since there is never any direct communication between the judge and employees, there is little reason to expect judge stringency to reverse for some employees and not others. We nevertheless show that all sub-sample estimates of the relationship between judge stringency and liquidation rates are positive and significant in Column (I) of Appendix Table A.1. To further gauge the monotonicity assumption, we break the instrument into the same subsamples, but redefine the instrument for each subsample to be the liquidation rate for all cases outside of the subsample. For example, for the low-wage workers, we use the judge's liquidation rate constructed from all cases except low-wage workers. Column (II) of Appendix Table A.1 reports the first stage estimates using this "reverse-sample instrument." While it may be a weak test of monotonicity, it shows that all estimates remain positive and statistically different from zero, suggesting that stricter judges for one type of case are also stricter for other case types.

4 Main Empirical Findings

4.1 Average Employment and Earnings Costs

We begin by examining the impacts of liquidating the firm on full-time employment, measured as being employed at the end of the calendar year. Panel (a) of Table 4 presents our 2SLS estimates and Figure 3a plots the point estimates together with 95 percent confidence intervals, illustrating that liquidation causes a statistically significant decline of 55 percent in the first year. By contrast, the average employment rate among employees in marginal firms (i.e., untreated mean) is slightly above 90 percent. The following year, the effect declines slightly to a 32 percentage point decline. After the first two years, the estimates are less precise but stable at around 7–8 percentage points.²¹ In Panel (b), we report estimates of the effect on any type of employment, including working at least 4 hours per week. We see that the estimates are similar in size.

estimate when negative individual effects are assigned negative weights. De Chaisemartin (2017) shows that researchers can identify a slightly differently weighted LATE when the assumption is violated as long as two equal proportions of defiers and compliers with identical potential outcomes exist.

²¹Appendix Table A.4 reports 2SLS estimates on unemployment insurance receipt in the last year of our observation window. We cannot reject that the effect is null. While we do not detect a statistically significant impact on disability insurance receipt, our evidence indicates that liquidation may increase withdrawal from the labor force and the likelihood of not having any labor earnings. We note that the estimates are imprecisely estimated, and we cannot reject a zero impact on labor force exit with conventional confidence levels. The impacts on labor force participation are not explained by early retirement eligibility at age 62. Excluding workers above 58 does not change the point estimates appreciably. These estimates are available upon request.

Figure 3: Labor Market Impacts of Employer Shutdown



Notes: This figure plots IV estimates. Panel 3a shows the evolution of the 2SLS estimates of impacts on employment and Panel 3b shows the corresponding estimates on wage income. The shaded blue spikes are 95 percent confidence intervals. See Table 4 for a description of the baseline sample and specification. Average exchange rate in 2024 is NOK/\$=10.

Years after decision (<i>t</i>):	0	1	2	3	4			
Panel (a):	Full Time Work							
Liquidation (β)	-0.537***	-0.319***	-0.131	-0.078	-0.074			
	(0.156)	(0.116)	(0.087)	(0.083)	(0.079)			
Overall mean of dependent variable	0.675	0.662	0.670	0.673	0.672			
Untreated mean	0.925	0.815	0.716	0.789	0.805			
Panel (b):	Part or Full Time Employment							
Liquidation (β)	-0.574***	-0.281**	-0.166*	-0.132*	-0.099			
	(0.156)	(0.109)	(0.089)	(0.077)	(0.078)			
Overall mean of dependent variable	0.704	0.708	0.723	0.733	0.731			
Untreated mean	0.981	0.838	0.779	0.826	0.865			
Panel (c):		Total Wage	e Income (N	OK1,000)				
Liquidation (β)	-125.971**	-122.341**	-104.519**	-111.192**	-71.045			
-	(49.269)	(58.839)	(51.004)	(55.318)	(51.543)			
Overall mean of dependent variable	429.722	403.118	402.357	406.839	407.947			
Untreated mean	485.238	441.915	438.853	454.817	450.598			
F-statistic	15.450	15.450	15.450	15.450	15.450			
Number of observations:	14,826	14,826	14,826	14,826	14,826			

Table 4: 2SLS Estimate:	Employment and Earnings

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. The sample includes firms with at least five employees before the petition is received and full-time employees aged 25–62. There are 312 judges. Each regression controls for fully interacted year and court indicators, a linear and quadratic of leave-out judge processing time, and individual controls for age, education, marital status, immigration indicators, and dummies for quintiles of pre-petition wage income levels. Employment is defined as having any wage income in a year. Appendix Table A.2 reports the first stage and reduced form effects of judge stringency. Average exchange rate in 2024 NOK/USD \approx 10.

Panel (c) of Table 4 reports our estimates of the effects of closing the firm on annual total gross wage. As we find important effects on any employment, reported in Panel (c), we include zeros for those without a job and plot the evolution of the 2SLS estimates in Figure 3b. The estimates show that annual wage income falls abruptly in the petition year, corresponding to a one-quarter decline in earnings relative to untreated employees. The earnings loss remains statistically significant and large for the following two years. By year four, however, the estimate is almost halved and is no longer statistically significant. Comparing the wage and employment effects, the relative impact on wages is larger than the relative fall in employment. This suggests the wage effect is driven by a combination of hours worked, hourly wages, and the extensive margin. This evidence is similar to, but less precisely estimated than Cederlöf (2020), who uses a quasi-experimental approach to study earnings losses from layoff in Sweden.

4.2 Employee Turnover

How many workers avoid the costs of layoff by moving to another firm? To assess the role of self-insurance from worker mobility, we need a measure of what workers know, how quickly they learn about distress, and how acting on information affects earnings in the next job. An important drawback with our data is that it does not provide exogenous variation in distress exposure. However, the information about petition dates allows us to examine the extent to which and how long information is available to the public using time-variation in changes to firm sales. We find that sales decline by 10 percent in the year before a petition is received, followed by an additional 30 percent drop in the same year the petition is received. Appendix Figure A.3a shows average sales relative to the petition date. Moreover, we find that a 10 percent decline in sales bears little signal of bankruptcy risk the following year, as illustrated by Appendix Figure A.3b. The increase in petitions at negative ten percent sales growth is barely noticeable.

To explore worker turnover around the petition date, we use the monthly employer-employee data to test the idea that abrupt changes in turnover before or at the petition date suggest that employees are learning about bankruptcy risk. We define a voluntary job-to-job (EE) move as a transition between employers from one month to another without an intervening spell of unemployment or non-employment for each individual starting 12 months before to up to three months after the petition date.

Next, we estimate the potential outcomes of *untreated* compliers and plot the estimated potential EE and unemployment spells around the petition date. As seen by the abrupt change in potential turnover rates, about 24 percent of all workers in marginal firms leave the firm before the judge has made a decision. While the confidence interval is wide, Figure 4 shows that turnover clearly increases in the month of the petition. In contrast, the employment-to-unemployment transition rate among the untreated is low and quite stable over the period. The increase in job-to-job moves prior to the liquidation is likely to attenuate the estimated earnings losses compared to a scenario where workers cannot anticipate future job loss.





Notes: The left figure shows the share of untreated workers in distressed firms with a job-to-job transition between 12 months prior to and 3 months after the bankruptcy petition date. The right figure shows the incidence of unemployment spells in the months around the bankruptcy petition. For job-to-job transition rates, the resignation time is imputed based on the average three month mutual dismissal time. Untreated means are potential outcomes of employees in marginal firms (see Dahl *et al.*, 2014).

4.3 **Outside Options**

To explore how changes in pre-layoff transitions, wage losses and outside options interact, we make use of information about local labor market conditions. To do so, we need to address the challenge of how to measure the local labor market. Berger *et al.* (2023) show that 4-digit occupations capture about 40 percent of transitions from one employer to another in the Norwegian labor market, and that this self-flow rate increases to 45 percent using 3-digit occupations. Moreover, the geographic mobility rate across commuting zones is quite low.²² These findings indicate that the combination of 3-digit level occupations in the commuting zone can serve as a reasonable proxy for the local labor market. We use the number of job openings divided by the number of registered job seekers within each local labor market as the tightness.

While we do not have experimental data on offer arrival and interview rates, we can alleviate some concerns about confounders across labor market conditions by residualize the tightness measure. In our main specification, we remove the covariance of tightness and permanent differences between occupations and commuting zones, as well as the female share and polynomials for av-

²²We examine whether liquidation affects the likelihood of moving to new local labor markets. While we find no statistically significant impact, we lack the statistical precision to draw firm conclusions. We use the geographic unit defined by commuting statistics (see, Berger *et al.*, 2023), and report our estimate in Panel (c) of Appendix Table A.5. Similarly, an influential line of research has established that switching careers is associated with larger earnings losses (see, e.g., Neal, 1995, Huckfeldt, 2022, and Braxton & Taska, 2023). While a career outside the firm may be difficult to define, we use the occupational code, a natural definition of the career ladder for several skilled occupations. For instance, the group "pharmacists" includes over-the-counter shop assistants, licensed occupations for handling prescriptions, and senior positions in private pharmacies and clinical and hospital pharmacists. We examine whether liquidation increases the sectoral mobility, defined as 4-digit occupation or industry, but find no statistically significant impact by the end of year 4. These results are reported in Appendix Table A.5.

erage age and tenure in the occupation and commuting zone. Moreover, combining the tightness measure with the linked employer-employee data allows us to distinguish between employees in the same firm who simultaneously experience strong and weak labor market conditions arising from belonging to different occupational groups (e.g., engineers vs. economists). After this first step, we divide the sample by the median vacancies per registered job seeker, and calculate the average vacancy-unemployment ratios in weak and strong markets, which are 0.245 and 0.798 respectively. In our sample, 43 percent of firms have workers in both strong and weak markets.



Figure 5: Heterogeneous Employment Impacts

Notes: This figure plots heterogeneity in impacts of shutting down a firm by the strength of the labor market the year of the petition. Strong and weak labor markets are defined by residualized vacancy-unemployment ratios in the prior 3-digit occupation and commuting zone. See variable, sample descriptions, and regression details in Table 4. The spikes are 95 percent confidence intervals.

We begin by exploring whether differences in outside options translate into differences in employment rates. If it does, differences in wage effects reflect differences in impacts on the extensive margin. To assess the extensive margin effects, we re-estimate equation 2 separately for the two samples using bi-monthly data. The outcome of interest is equal to one if the worker has any employment by the end of the month, and Figure 5 shows no clear evidence of differential impacts, suggesting that the availability of jobs does not affect the likelihood of being employed.

Next, we investigate whether differences in job opportunities lead workers to accept jobs with lower wages or shorter hours. Our focus is on the average annual earnings and the pre-bankruptcy job mobility rates, and we report our headline results in the second columns of Panels (a) and (b) of Table 5. The 2SLS estimates show that earnings losses are 21 percent in weak markets and 9 percent in strong markets but we lack the statistical precision to firmly conclude that the estimates are different from each other. Turning to the employee reallocation rates, we find that turnover is

three times higher in strong labor markets than in weak ones.

	(a) W	'eak Labor N	Aarket	(b) Strong Labor Market			
	Firm	Wage	Pre-layoff	Firm	Wage	Pre-layoff	
	Closes	Income	turnover	Closes	Income	turnover	
	(1)	(2)	(3)	(1)	(2)	(3)	
Judge Stringency	0.911***	-91.402**		0.661***	-22.902		
· · ·	(0.234)	(45.018)		(0.209)	(52.146)		
Liquidation (β)		-100.303**			-34.665		
1		(47.507)			(78.006)		
Overall mean	0.502	390.572		0.610	431.996		
Untreated mean		475.263	0.137		381.057	0.346	
Observations:		7,861			6,945		

Table 5: 2SLS Estimates: By Labor Market Conditions

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the first stage, reduced form, and second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. See variable, sample descriptions, and regression details in Table 4. Strong and weak labor markets are defined by residualized vacancy-unemployment ratios in the prior 3-digit occupation and commuting zone. When residualizing year-of-petition vacancy-rates, we remove the covariance between gender, commuting zone, and occupation-fixed effects and a quadratic function of age and tenure.

To assess the strength of our conclusions about outside options, we bootstrap the sample with 200 repetitions and estimate a bootstrapped standard error of the differences in potential turnover rates. In our baseline specification, we find that the difference equals 0.111, allowing us to weakly reject the null hypothesis of equal means. We estimate two other specifications, where we omit (a) worker controls, and (b) the dummies for commuting zones in another specification and report estimates in Appendix Table A.6. The results remain quantitatively similar, where the differences in estimated turnover rates are significant at conventional levels.²³ As an additional robustness check, we split the three tightness measures into quintiles and re-estimate the wage loss and turnover rates for each of these 15 subsamples. In Figure A.4, we plot the association between turnover and wage losses, finding it is positive, indicating that turnover induced by more job opportunities may reduce the size of estimated earnings loss from 25 to 5 percent.

4.4 Heterogeneity

Next, we explore differences in estimated impacts across additional splits of our data. Our outcomes of interest are the average earnings from when the petition is filed to four years after, and the average anticipatory turnover among untreated workers in marginal firms.

We begin by splitting the sample by firms that pay above median estimated premiums, and by workers that are high- vs. low wage according to the Abowd *et al.* (1999) model of log wages

²³We bootstrap the sample with replacement 200 times, and estimate the standard error of the differences in potential turnover rates. In both specifications, the probability of falsely rejecting the null is less than 1 percent. The estimated difference in impact on average earnings is not statistically significant at conventional levels, but yield p-values of 20 to 30 percent.

and two-way fixed effects. Table 6 reports estimates of the first stage, the reduced form effects of judge stringency, and the 2SLS estimate of each sub-sample. Panels A and B show that while our estimated five-year earnings losses are similar, that the out-migration rates are much larger for workers in the high-wage firms. This finding is consistent with workers with more to lose from layoff searching more intensely, offsetting some of the earnings loss. In Panels C and D, we see that high-wage workers suffer larger earnings losses than low-wage workers. These estimates are statistically significant different from each other with a p-value of 0.028.

We further split these subsamples by interacting high- and low wage-firms and workers, and plot the estimates in Figure 6. The figure illustrates that higher employee reallocation rates seem to reduce the five-year employee costs. In contrast, Appendix Figure A.4 shows a flat relationship between earnings losses and pre-bankruptcy unemployment risk. The evidence points toward labor market insurance from job search, rather than differential scarring from unemployment.

In addition, we explore further sample splits, including age and skill level, and report estimates in Appendix Table A.7. We see that five-year earnings losses are concentrated among workers above 40, potentially indicating that older workers have weaker incentives to invest in a new career because they have fewer years remaining until retirement. In contrast, we find that earnings losses are quite similar for high- and low skilled, measured as having completed high-school or not. Somewhat surprisingly, there are no clear differences in reallocation rates between high vs. low skill or young vs. older workers. Rather, differences in turnover seem to be driven by differences in labor market conditions and whether a firm is high or low-wage.

	A. High-V	Wage Firm	B. Low-Wage Firm		C. High-Wage Worker		D. Low-Wage Worker	
	Firm	Wage	Firm	Wage	Firm	Wage	Firm	Wage
	Closes	Income	Closes	Income	Closes	Income	Closes	Income
	(1)	(3)	(1)	(3)	(1)	(3)	(1)	(3)
Judge Stringency	0.985***	-74.834	0.578**	-60.472	0.687***	-137.989**	0.911***	-39.465
	(0.337)	(68.315)	(0.254)	(44.590)	(0.234)	(58.821)	(0.199)	(33.245)
Liquidation (β)		-75.970		-104.651		-200.978**		-43.315
		(64.692)		(89.757)		(92.792)		(36.233)
Overall mean	0.560	464.397	0.553	356.426	0.571	451.824	0.533	366.277
Untreated mean		489.045		347.771		537.195		395.922
Pre-layoff turnover		0.259		0.033		0.190		0.296
Observations:	7,8	801	5,	967	7	,241	7,	801

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the first stage, reduced form, and second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. Untreated means are potential outcome of employees in marginal firms (see Dahl *et al.*, 2014). The sample includes firms with at least five employees the year before the petition. The estimation sample includes full-time employees aged 25–62 employed 12 months prior to petition. There are 312 judges. Each regression controls for fully interacted year and court indicators and includes the individual demographic control variables from Table 2 and dummies for each quintile of prepetition wages. The mean wage income is winsorized at the 99th percentile and disposable income is winsorized at the 95th percentile. Outcomes are averaged over the five-year period and are deflated to 2015-values using consumer price index. Average exchange rate in 2024 is NOK/\$=10.

Figure 6: Heterogeneous Wage Impacts by Pre-Petition Worker Turnover Rates



Notes: This figure plots heterogeneity in impacts of shutting down a firm on wage income against the fraction of untreated workers who leave their employer before or during the month of the petition. See variable, sample descriptions, and regression details in Table 4. The size of scatter plots reflect the inverse of the standard errors of the estimated impact on wage income.

4.5 Private and Public Insurance

To contrast the variation in earnings losses across labor market conditions, we investigate the role of the tax and transfer system and self-insurance from private assets. Starting with taxes and unemployment insurance, Table 7 reports the five-year impact estimates, illustrating that the impact on UI benefits is about 11 thousand, corresponding to a 130 percent increase relative to workers in non-liquidated firms. The third column reports the estimated impact on five-year average disposable income, which accounts for capital and business income, government transfers such as social assistance (i.e., welfare), housing, unemployment, and disability insurance benefits, all net of taxes. The average impact over the period is equal to 53 thousand—about half of the earnings loss, but still a sizable loss in income. Appendix Figure A.6 illustrates that the tax system automatically buffers the impact on disposable income over the entire period, whereas the UI system plays an important role, but only in the first two years after the petition has been sent.²⁴

These estimates may exaggerate the costs of layoffs if other family members can increase labor supply (e.g., Cullen & Gruber, 2000). We examine insurance from within the household from added workers (i.e., increasing work from a spouse). The third and fourth columns of Table 7

²⁴In Panel (b) of Appendix Table A.9, we examine the role of other market income sources, including business and capital income. We do not reject the null hypothesis that liquidation has no effect on other sources of market income. We cannot detect any statistically significant effect on any of the other sources of transfer payments.

show how liquidating the firm affects individual and household disposable income. While adding spousal income reinforces the decline, we cannot offer a definitive conclusion due to a lack of statistical precision.²⁵ Households can also smooth consumption by pooling assets. We estimate the impacts of liquidating the firm on net assets at the individual and household levels; columns (5) and (6) show statistically significant effects corresponding to a relative decline in net wealth of about 60 percent of the cumulated income loss. Again, we do not see any statistically significant differences between the impacts on individual and family-level assets, possibly reflecting the low marriage rates and added noise from changes in the composition of marriage rates over time. Appendix Table A.9 shows that the impact on net wealth is quantitatively similar if we only focus on the last year and if we adjust for passive savings from housing market appreciation.

	Wage	UI	Disposable	Add Spouse	Δ	Add Spousal
	Income	Benefits	Income	Income	Net Assets	Assets
	(1)	(2)	(3)	(4)	(5)	(6)
Liquidation (β)	-107.014**	10.529*	-52.557**	-69.839**	-247.460**	-237.064*
	(48.702)	(5.582)	(22.859)	(34.250)	(116.132)	(133.039)
Overall mean	409.996	15.785	340.647	434.663	242.420	347.502
Untreated mean	454.284	8.956	354.026	457.273	451.661	569.728
Number of observations:	14,826	14,826	14,826	14,826	14,826	14,826

Table 7: 2SLS Estimates of Impacts on Disposable Income and Assets

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. Untreated means are potential outcome of employees in marginal firms (see Dahl *et al.*, 2014). See variable, sample descriptions, and regression details in Table 4. Outcomes are averaged over the five-year period and are deflated to 2015-values using consumer price index. Average exchange rate in 2024 is NOK/USD ≈ 10 .

Previous research has documented that laid off workers experience higher personal bankruptcy rates (Keys, 2018), and that credit conditions deteriorate among laid off low-wage workers (Aaronson *et al.*, 2019). Braxton *et al.* (2020) show that access to (pre-established) lines of credit is an important insurance mechanism for job loss in the US. We examine these channels in the Norwegian setting and report our 2SLS estimates in Appendix Table A.10. Our findings suggest that the decline in net wealth reflects the use of personal assets and not debt increases. While somewhat imprecise, our 95 percent confidence interval rejects debt increases by more than half of an annual salary over five years after the firm's liquidation. We do not find clear evidence that liquidation worsens access to credit markets, proxied by the debt relief from personal bankruptcy. Furthermore, we estimate the impacts of liquidation on an indicator equal to one if the person's wage is garnished at some point after the petition date and the average interest rate paid on debt. We cannot reject the null hypothesis of zero impact on either of these outcomes.

²⁵A natural question is whether losing a valuable employment relationship affects marriage rates. While Appendix Table A.3 shows that we do not detect any statistically significant effect on marriage rates, marriage rates are low at the beginning of the period and steadily increase over time. This is why, instead of conditioning the sample on marital status, we include the spousal income when we observe a registered spouse in the data.

4.6 Household Consumption and Public Expenditure

Our last exercise explores the impacts of liquidation on consumption expenditure and net government transfers. To impute consumption expenditure, we apply the accounting identity, which states that the effect on consumption, ΔC must equal the effect on the change in net wealth ΔW plus disposable income ΔI (see, e.g., Browning & Leth-Petersen, 2003 and Eika *et al.*, 2020). Because the market valuation of housing is only available from 2010, our target parameter is the effect of employer shutdown on the five-year average consumption expenditure.²⁶ Column (1) of Table 8 copies the impact on disposable income, and column (2) includes the impact on active dissaving. Column (3) reports the average impact on annualized imputed consumption expenditure. Eika *et al.* (2020) estimate the value of owner-occupied housing relative to the rental cost and find the rent-to-value ratio for 1994–2014 is 2.88 percent. Column (4) shows that adding the consumption value of owner-occupied housing reduces the impact on annualized consumption by about 16 thousand.

To explore the fiscal cost of liquidating a firm, we report our estimated impact on the sum of all transfers minus taxes in last column of Table 8. This point estimate shows that a liquidation causes a net increase in public expenditure of 63 thousand NOK per year— equal to NOK315,000 over the first five years, or, about sixty percent of the pre-bankruptcy annual gross earnings.²⁷

Years after decision (<i>t</i>): average 0–4, NOK1,000 = USD100								
	Disposable	Active	Imputed	Imputed Consumption				
	Income	Dissaving	Baseline	Add Housing	Cost			
	(1)	(2)	(3)	(4)	(5)			
Liquidation (β)	-52.557**	-207.374*	-8.710	-24.988	63.429***			
	(22.859)	(115.727)	(28.433)	(28.784)	(22.974)			
Overall mean	340.647	332.833	285.083	318.877	-72.420			
Untreated mean	354.026	324.445	297.553	339.089	-117.570			
Number of observations:	14,826	14,826	14,826	14,826	14,826			

Table 8: 2SLS Estimates of Impacts on Imputed Consumption and Fiscal Costs

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. Untreated means are potential outcome of employees in marginal firms (see Dahl *et al.*, 2014). See variable, sample descriptions, and regression details in Table 4. Fiscal costs include all transfers from the government, less taxes paid, where transfers and taxes are windsorized at the 99th percentile. Disposable income, assets, and imputed consumption are winsorized at the 95th percentile. Disposable income is averaged over the five-year period, the other outcomes are cumulative outcomes, measured four years after petition. All outcomes are deflated to 2015-values using consumer price index. Average exchange rate in 2024 is NOK/USD ≈ 10 .

²⁶The four-year average uses the last observation for housing, which is available for the entire sample. An advantage of using the cumulative change in income and wealth is that noise from within-year changes in wealth due to portfolio rebalancing (i.e., from selling stock or housing market transactions) is averaged out. We examine the effects of employer shutdown on other types of capital income, including interest rates, sales of listed and unlisted stocks, and realized gains from selling real estate in Appendix Table A.9. Reassuringly, we find no evidence of capital income effects nor do we detect statistically significant effects on business income.

²⁷These costs are relatively small compared to the average debt per employee among larger financially distressed employers in our sample, which amounts to NOK1.5 million (approx. \$150,000).

4.7 Internal and External Validity

Before we explore the welfare implications of our findings, we discuss the internal and external validity of our findings. We report the estimates of wage income and disposable income effects of liquidation in Appendix Table A.11, and summarize the output from the exercises here.

To explore the internal validity, we start by excluding judges who handle fewer than 30 cases. This allows us to assess the extent that our findings are sensitive to noise in our judge stringency measure. Second, we address a concern that the relationship between judge stringency and individual outcomes is non-linear, where weights for subgroups of compliers with heterogeneous treatment effects would be misspecified. To assess such potential misspecifications, we implement the IV model using a quadratic and cubic specification of judge stringency. Third, we allow for a more flexible specification of judge processing time. Reassuringly, our conclusions remain unchanged across these three specification checks. Lastly, we show that average impacts are similar but much less precisely estimated when we include numeric values above the 99th percentile.

In terms of external validity, we follow the empirical specifications in Lachowska *et al.* (2020) and show that mass layoffs lead to similar earnings losses in Norway as in the US. These results are presented in Appendix C.

5 A Job Ladder Model With Learning and Self-Insurance

To quantify the value of a specific employer and how information affects reallocation, we develop a model that can match our key empirical facts.

5.1 Environment

The model runs in discrete time, starting from the month a worker enters the labor force at age 25 until the month the worker turns 67 and retires. Workers search for jobs while employed and unemployed, and decide how much to save for retirement or to offset cuts in consumption upon job loss. We let $c_t \ge 0$ denote consumption in month *t*. Individuals maximize

$$\mathbb{E}_{0}\left[\sum_{t=1}^{T}\beta^{t-1}\left(u(c_{t})-\psi(e_{t})\right)+\beta^{T}\nu(a_{T+1})\right]$$
(3)

subject to the constraints we specify below. Workers receive new job offers by exerting search effort, denoted e_t , where $\psi(e_t)$ is the corresponding disutility of effort. The net value of all assets held by the worker at the end of the *T*-period working life is denoted by $a_{T+1} \ge \underline{a}$ and we let $\nu(a_{T+1})$ capture the utility of bringing assets into retirement. Following Lise (2013), we assume that within-period consumption preferences are given by

$$u(c_t) = (c_t^{1-\gamma} - 1)/(1-\gamma),$$
(4)

where γ is the degree of relative risk aversion. Moreover, search effort incurs a disutility following the power form,

$$\psi(e_t) = \mu \frac{e_t^{\eta}}{\eta} \tag{5}$$

where $\eta > 1$ is the elasticity of search costs with respect to effort and μ is a scaling parameter. We further assume that residual utility gained from net assets held at the end of the final working period *T* is given by

$$\nu(a_{T+1}) = \varphi \frac{(a_{T+1} - \underline{a})^{1-\gamma} - 1}{1 - \gamma}$$
(6)

where φ is a scaling parameter and <u>a</u> is the borrowing limit at the time of retirement. This residual utility term captures both the benefits the worker enjoys from carrying assets to be consumed during retirement as well as any warm-glow utility the worker obtains by leaving bequests at the end of the life-cycle. We present the workers' value function in full detail in Appendix D.1.

Search and Income Process A frictional matching process determines the per-period income of employed workers. The worker receives a constant pre-tax wage w while employed, but can search for an employer offering a higher wage in each period. Workers determine the likelihood of receiving an employment offer, $\lambda_{i,d,M}e_t$, by choosing effort $e_t \in [0, 1]$. The maximum offer rate $\lambda_{i,d,M}$ varies with the labor market condition, M, and whether or not the worker's current firm is undergoing bankruptcy proceedings, d. This captures differences in the labor market tightness that workers face. Motivated by the findings in Faberman *et al.* (2022), we allow the offer rate to vary with employment status, i, reflecting differences in search efficiency on-the-job and during unemployment. Employment offers are drawn from a constant wage distribution G(w|k, M), that may vary by market condition M and worker's skill type k (a "low" or "high" type, corresponding to the skill categories defined in Section 4.4).

Employed workers' earnings are a function of the pre-tax wage w, supplemented by stochastically increasing human capital, h. In particular, there are a finite number of observable human capital types $h \in \{0, h_1, ..., h_n\}$ and workers are paid $w \cdot exp(h)$ before taxes each period. Human capital decreases stochastically during spells of unemployment, representing the degradation of skills and allowing the model to match the size of earnings loss when shutting down the firm that we observe empirically. Each period, employed workers gain one level of human capital with probability Δ_h^+ and unemployed workers lose one level of human capital with probability Δ_h^- . Workers observe their new level of human capital at the beginning of each period, prior to making any decisions.

Workers are exogenously separated from their employers with probability $\delta_{k,M}$ each period, which may vary by market conditions and the worker's skill level. Experiencing this separation shock means a worker enters unemployment the next period. Unemployed workers receive a per period benefit which depends on their labor force participation and earnings in the most recent

period of employment, (\tilde{w}, \tilde{h}) . Workers who are actively searching for employment are eligible for an unemployment insurance (UI) benefit which is proportional to past earnings $\tilde{w} \cdot exp(\tilde{h})$, up to some maximum level \bar{b} . The proportional UI benefit system generates voluntary unemployment in the sense that workers may reject some employment offers. Workers who do not actively search for employment are not eligible to collect UI benefits, but are instead eligible for a basic social assistance program which provides non-participants in the labor force with a minimum level of income, denoted by \underline{b} .

Workers face a variety of taxes which are not explicitly modeled. Instead, we follow Heathcote *et al.* (2014) in approximating the net of tax income by $\omega_1 \left[i \cdot (w \cdot exp(h)) + (1-i)b(\tilde{w}, \tilde{h}, e)\right]^{\omega_2}$, where ω_1 and ω_2 capture the level and progressivity of taxes in our setting, respectively. Post-tax income can be used for consumption, and individuals can save and borrow through a one-period bond with a return normalized to one. We allow the bond price to differ when saving versus borrowing to reflect potential differences in interest rates and we let q_b and q_s express the per-unit bond price when borrowing and saving, respectively. Workers face a natural borrowing limit as the result of the constraint that consumption in each period is non-negative and that assets held at the time of retirement must exceed the borrowing limit \underline{a} .

Learning about Distress. On top of the standard separation risk, workers can experience a distress shock—representing the realization or anticipation of a bankruptcy petition—with a potential liquidation following. The distress shock occurs with probability $\pi_{k,M}$, varying by the worker's skill type and the labor market conditions. Importantly, distress means an individual updates her expectation of her employment being terminated, lowering the value of the current position. We assume that firms which enter distress do not exit distress without proceeding through the bankruptcy courts. In some period after the onset of distress, a formal petition is filed and one period later a bankruptcy judge decides to liquidate the firm or dismiss the case. Some workers learn that their firm is in distress before their employer is formally petitioned and workers who anticipate a potential liquidation revise their beliefs about its likelihood over time. Other workers may not fully internalize the increased risk of their position being terminated even after a formal petition has been filed against their employer.

We model workers' information and beliefs to capture both of these properties. The share of workers who first realize their job is at risk due to bankruptcy *n* periods before the judge's decision is made is denoted ξ_n^M . Assuming a maximum of three periods of advance information, a worker who becomes aware that her firm is in distress does not know when the judge's decision will be made, but knows the distribution of workers becoming aware $\{\xi_n^M\}_{n=1}^3$. With this knowledge, the worker can form Bayesian beliefs about the probability that the judge makes a decision in each period. For example, if the distribution of advance information were defined by $\{0.5, 0.3, 0.2\}$, then in the first period the worker expects the judge to make a decision at the end of the current period with probability 0.5. If the judge's decision does not occur at the end of the first period, she updates her belief because she knows she is not part of the 0.5 share of workers who receive one

period of information. In the second period, she believes she will face a potential liquidation at the end of the period with probability $\frac{0.3}{1-0.5} = 0.6$. Appendix D.2 provides a full example of how worker beliefs update over time as a firm progresses through bankruptcy proceedings.

5.2 Timing

In the period where firms enter distress, all workers are aware that a bankruptcy case is assigned to one of two types of judges. With probability \overline{L} , the case is assigned to a strict judge who always liquidates the firm; with probability $1 - \overline{L}$, the case is assigned to a lenient judge who always dismisses the case. The judge makes a decision at the end of the period. When the case is dismissed, the distress shock is resolved, with no long-term effects for the worker.²⁸ When the case is not dismissed, the worker's employment is immediately terminated and the worker enters the next period unemployed.

Figure 7 illustrates the timing of the firm-bankruptcy process together with workers' advance information. The judge's decision and the subsequent liquidation are timed such that workers make their search, consumption, and savings decisions after learning if their firm is distressed but before learning how the distress shock will be resolved. Thus, workers in distressed firms face heightened earnings risk. As a result, workers in distressed firms will have lower reservation wages when searching on the job and may exert greater search effort or increase their precautionary savings as a form of self insurance.



Figure 7: Timeline of Firm Distress Proceedings

Notes: This figure shows the timing of the bankruptcy process and workers' advance information of the process in our model. We condition the sample on workers being in the distressed firm twelve months prior to the bankruptcy petition being filed, and we allow distressed firms to face a formal bankruptcy petition in any month of year 0. We measure outcomes from the beginning of year 0 through the end of year 4. When we refer to the sub-sample of workers in "strong" and "weak" market conditions, we refer to the market conditions *M* that the worker is experiencing at the beginning of year 0.

6 Bringing the Model to the Data

This section discusses the estimation of parameters and the fit of our estimated model.

²⁸Workers in distressed firms still face the baseline probability $\delta_{k,M}$ of separating from their firms even if liquidation does not occur, which we treat as the worker choosing to leave for other non-wage reasons.

6.1 Samples and Market Conditions

In the model, as in the empirical setting, workers are either treated (i.e., their employer is liquidated) or untreated (i.e., their employer experienced distress but not liquidation), even if they leave the distressed firm before the judge's decision is made.

The overall distress rate and labor market flows can vary by labor market conditions M and we allow for heterogeneity in the effect by skill type (changes in $\delta_{k,M}$ and $\pi_{k,M}$). In addition, both skill types face lower probabilities of receiving an employment offer conditional on effort (a decrease in $\lambda_{0,d,M}$ and $\lambda_{1,d,M}$) and we allow for a different distribution of advance information { ξ_n^M } across strong and weak labor markets. Consistent with our empirical design, when we refer to the subsample of workers in strong and weak labor markets, we refer to the market conditions at the start of year 0 (i.e., the year in which the judge's decision occurs).

6.2 Estimation

We borrow some parameters from the literature, such as the coefficient of relative risk aversion. We use $\gamma = 1.7$, which is consistent with the median value reported by Chiappori & Paiella (2011). The subjective discount factor is set to to $\beta = 0.9975$, implying an annual subjective discounting rate of approximately three percent, which is in line with common estimates from the literature.

External estimation. Other parameters are derived from the institutional setting. The replacement rate, r = 0.624, and maximum unemployment insurance benefit, $\bar{b} = 28.080$ NOK1,000, are set in accordance with the Norwegian UI system, and the minimum income provided by social assistance programs for workers who are unemployed and not actively searching is set to $\underline{b} = 8.333$ NOK1,000 monthly. We obtain estimates $\omega_1 = 5.336$ and $\omega_2 = 0.844$ from the 2015 tax system.²⁹ The price of the one-period bond when saving is set to $q_s = 1$ consistent with zero real interest rate on savings and we set $q_b = 0.9925$ to be consistent with an annual real interest rate of 9.4% on borrowing which is in line with typically observed rates in our setting. We set the borrowing limit at retirement to $\underline{a} = -593.617$ NOK1,000 which is the (negative of) net present value of the lowest retirement pension income available in our setting over a horizon based on the typical length of retirement.

We estimate some parameters from auxiliary data sources. The transition between strong and weak market conditions follows a Markov process, with probability τ_M of remaining in state M. We set $\tau_0 = 0.9788$ so that the expected duration between spells of weak market conditions is 47 months, and we set $\tau_1 = 0.9612$ so that the expected duration of weak market conditions is 26 months, consistent with the average length of local labor market conditions in our sample. The distribution of wage offers is estimated from data on the joint distribution of wages, vacancies, and

²⁹This procedure is implemented in two steps. First, we simulate the actual tax system in 2015 for values of gross wage income ranging from 100 to 1,500 thousand NOK. Second, we regress the log of simulated disposable income on the log of gross wage income to obtain the parameters ω_1 and ω_2 .

job seekers, where we first obtain residualized and recentered wage and vacancy-unemployment ratios. These control for dummies for each labor market region and occupation and a flexible function of age. We then compute the average wage in each occupation-region cell and use the labor market tightness as weight, and we perform this calculation separately for each skill type and labor market condition. We find that for all skill type and market condition combinations, the distribution of accepted offers is approximately log-normal and we calculate the CDF of residualized tightness and average wages. The mean and standard deviations of the resulting CDFs are the parameters for the log-normal offer distribution *G* for skill type *k* in market condition *M*.

The aggregate employment transition rates are calculated using overall population statistics and sample weights. The separation probabilities $\delta_{k,M}$ for strong and weak labor market conditions are taken from the data and combined with the share of each skill type in our overall and unemployed samples, as reported in Table 1, to estimate conditional probabilities by skill type. Similarly, we calculate distress risk, $\pi_{k,M}$, by combining the approximate share of firms entering distress each year in each market condition with sample shares of high and low-skilled to determine how the risks differ by skill type. Conditional on the firm being distressed, we set the probability of being assigned to a strict judge (i.e., *always liquidate*) to $\bar{L} = 0.553$, equal to the mean liquidation rate from Table 3. We impose constraints on the offer arrival rate based on the observed market tightness, but allow the exact rates $\lambda_{i,d,M}$ to be calibrated internally. In particular, for each employment status *i*, we fix the ratio of $\lambda_{i,d,M}$ values from the residualized ratio of vacancies to unemployed workers in each local labor market described in Section 4.3. Next, we use the recentered average ratios and combine them with the sample of workers in distress to measure the tightness by distress status.

Internal estimation. The remaining 13 parameters are estimated internally: the strength of the retirement savings motive (φ), the elasticity and scale of search costs (η , μ), the maximum offer arrival rates (λ_i), the distribution of advance information about distress ($\xi_1^M, \xi_2^M, \xi_3^M$), and the human capital process (Δ_h^+, Δ_h^-). We directly estimate 11 of these parameters with 12 moments to target.³⁰ These parameters are estimated using the Method of Simulated Moments (MSM), which finds parameter values that minimize the distance between simulated moments and some of our key reduced form evidence. For each candidate set of parameter values, we solve the worker's problem by backward induction and simulate a cohort of workers from age 25 to age 67. Workers experience a local business cycle, changing the aggregate state of unemployment risk, liquidation risk, and employment offer arrival rates, lasting 26 months and happening with a 47 month spacing over the simulated life-cycle.³¹

³⁰While there are 13 total parameters remaining in the model, we only need to calibrate 11 of these parameters, as the value of ξ_1^M in each market condition *M* is implied by the values chosen for ξ_2^M and ξ_3^M along with our assumption that workers receive only up to three periods of advance notice about potential liquidations. In this sense, the model is over-identified despite having 12 targeted moments and 13 parameters that are not set externally.

³¹To ensure that the share of observations from each market condition in our simulated data match those in our empirical data when calculating moments, we take weighted averages for population-level calculations and randomly select a properly-balanced sub-sample for simulated regressions.

Workers who learn that their firm is in distress become part of our simulated "quasi-experimental sample." To mimic the sample restrictions in the data, we condition our simulated bankruptcy sample of workers on having twelve months of potential tenure at the firm at the time of the judge's decision and being aged 62 or below so we can observe simulated outcomes at least four years after the petition.³² Workers whose firm is liquidated—including those who leave the firm before the judge makes the decision—comprise the treated workers in our simulated sample, with the remainder of workers who experience distress being untreated.

Moments replicating 2SLS estimates from Section 4 are weighted according to their t-statistics, while moments that refer to the overall population receive a weight of one. Because worker reallocation away from distressed firms is central to understanding the costs of layoff, we assign a weight of ten to the reallocation rates in each local market state in order to ensure that the model replicates the rates found in our empirical evidence. The full list of targeted moments and their respective weights can be found in Appendix Table D.2.

Identification. Although we have no proof of identification, we provide a discussion of how different moments provide information that aid in estimating the parameters of interest.

There are four key parameters determining search behavior in the model. We start by considering how the elasticity of search costs with respect to effort, η , and the scale of search costs, μ , affect the worker's choice of effort. A higher search effort in response to layoff risk, implies a lower risk of experiencing a layoff and a faster recovery of their income, suggesting that the effect of liquidation on wage income in year 1 helps identify the elasticity η . Similarly, the higher the scale of search costs, μ , the less search effort workers will exert, and thus the longer unemployment spells will last. Thus, the effect of liquidation on UI benefit income in year 4 pins down the scale of search costs μ . Holding a worker's search behavior fixed, the maximum offer arrival rates affects employee transitions from unemployment to employment, λ_u , and from one employer to another, λ_e .

The distribution of advance information, captured by the parameters $\{\xi_n^M\}_{n=1}^3$, determines workers' ability to respond to a pending bankruptcy petition. Greater access to advance information means more time to search for a new employer and, thus, higher likelihood of making an EE transition before the judge's decision. Moreover, the beliefs induced by the distribution of advance information play an important role in determining workers' reservation wages. We argue that the combination of the share of workers leaving their firm prior to the bankruptcy judge's decision in each market state and the average loss of wage income from year 0 to year 4 in each state help pin down the distribution of beliefs.

The rate at which workers gain human capital while employed, Δ_h^+ , determines how quickly workers can recover their prior level of earnings following a layoff. Thus, the rate of human capital gain, Δ_h^+ , is determined by the effect of liquidation on wage income in year 4. Similarly, the

³²Because distress is, by construction, randomly assigned to workers in the model, we do not need to rely on the 2SLS approach and instead use the analogous OLS regressions.

rate at which workers lose human capital while unemployed, Δ_h^- , determines the size of persistent earnings loss following a layoff, with more rapid loss of human capital making prolonged unemployment more costly. Thus, the rate of human capital loss, Δ_h^- , is identified by the effect of liquidation on employment in year 4.

Finally, the residual utility scaling parameter, φ , determines workers' savings behavior near retirement, with larger values of φ encouraging greater savings. Targeting the ratio of average wealth of workers at age 67 to average wealth of workers at age 64 allows the model to approximate the savings behavior we observe empirically. In particular, the scaling of the residual utility term $\nu(a_{T+1})$ prevents workers from excessively dissaving as they approach the end of the model's finite horizon.

6.3 Results and Fit

We report the full set of calibrated parameters and simulated moment values in Appendix Table D.2. The model successfully replicates our 2SLS estimates. The model generates simulated estimates that fall within the 95% confidence interval around our empirical estimates of the impacts on employment in year 4, wage income and UI benefit income in both year 1 and year 4, and the average wage income loss from year 0 to year 4 in each local market condition. In terms of population moments, the fitted model generates slightly lower UE transition rates and slightly higher EE transition rates than what we observe in the data, as well as slightly higher savings near the age of retirement.

In addition to our targeted moments, we verify that the model generates reasonable estimates of several non-targeted moments. We calculate the average effect of liquidation on disposable income in year 0 through year 4 and report the results in Appendix Table A.12. We find that the average effect of liquidation on disposable income in the model is approximately 42 percent of the effect of liquidation on wage income, which is roughly consistent with our finding in Table 7, and suggests that the UI benefit system in the model offsets a similar portion of the loss in wage income as in the data. Moreover, we find that asset decumulation offsets an additional 43 percent of the loss in wage income, leaving a consumption expenditure pass-through of around 15 percent in the model and an average effect of liquidation on consumption of approximately -NOK27,700, both of which are within the 95% confidence interval around our estimates from Table 8. The overall asset decumulation due to liquidation is approximately 57 percent of the untreated mean, which matches our finding from Table 7, and suggests that the model does a reasonable job of matching the use of savings as an insurance mechanism when facing potential loss of employment. Finally, we use simulated data to estimate the average unemployment duration at pre-unemployment wages near the kink in the UI benefit schedule. We estimate an elasticity of unemployment duration with respect to benefits close to 1, which is in line with the literature using a regression kink design (RKD) to estimate the elasticity of unemployment duration with respect to benefits (see, e.g., Schmieder & Von Wachter (2016)). We also evaluate the model by performing a wage decomposition in Appendix Figure D.2, showing that human capital and search play an

important role in shaping the earnings recovery.

7 Assessment

This section uses the estimated model to understand how worker reallocation shapes the welfare costs of financial distress.

7.1 The Value of an Employer

We begin by quantifying the value of an employer. To do so, we calculate the employees' willingness to pay to keep their current job (i.e., avoiding liquidation). This quantity is defined as the amount of assets the worker would be willing to forgo at the moment the judge makes her decision to have the case dismissed. Workers who leave the distressed firm prior to the bankruptcy judge's decision have a WTP of zero by definition. For everyone who remains in the distressed firm until the decision occurs, we calculate an asset differential Δ_a satisfying the indifference condition

$$V_{\hat{t}+1}^{e}\left(a - \Delta_{a}, h, w, d_{0}, k, M\right) = V_{\hat{t}+1}^{u}\left(a, h, b(w, h), d_{0}, k, M\right)$$
(7)

where V_t^i denotes the value of each employment status $i \in u, e; d_0$ indicates being in the nondistressed state; and \hat{t} denotes the worker's age when the judge's decision is made. The lefthand side of the indifference condition denotes the worker's value if the case is dismissed and the worker's net wealth is reduced by Δ_a . The right-hand side of the condition denotes the worker's value at the moment the job is terminated—capturing the potential loss of human capital during the subsequent unemployment spell plus the expected utility cost of finding a new job and spending savings now rather than later, net of the unemployment benefits—assuming people are aware of the outside options in their local labor markets.

Scenario:	I. Baseline	II. Outside Options		III. Ideal Experiment	
		Strong Market	Weak Market	No Advance Information	
Willingness to Pay	439.312	407.740	468.149	502.151	
NPV of Income Loss	-445.967	-430.651	-463.852	-475.861	
Share Remaining at Firm	0.735	0.624	0.833	1.000	
WTP Conditional on Remaining	524.622	653.430	562.003	502.151	

Table 9: The Value of an Employer and Selection on Future Layoff Costs

Notes: This table reports the value of keeping the current employer to workers in distressed firms. We report the average amount of liquid assets that the worker would be willing to forgo to have their firm's bankruptcy case dismissed by the judge as well as the average net present value of the simulated impact on disposable income over the remaining life-cycle. The first column reports values from our baseline fitted model. The second and third columns report the estimated values among the sub-samples of workers who face strong and weak local labor market conditions, respectively. The fourth column reports the estimates in a counterfactual model where workers receive no advance information about the pending liquidation and instead immediately face layoff.
On average, we calculate that the WTP is about 439,000 NOK. We benchmark the average WTP to the net present value (NPV) of the impact of liquidation on disposable income over the remaining life-cycle. To compute these present values, we calculate the average effect of liquidation on the simulated workers' disposable income \tilde{t} periods after the judge's decision. We then compute the average age of workers at the time of the judge's decision, and compute the NPV as the discounted sum of the effects of liquidation each period of the average worker's horizon. We find that the average WTP equals approximately the NPV of the income loss at the time a judge makes her decision.³³ These calculations are reported in Panel I of Table 9, together with the share of worker's who remain at the distressed firm at the time of the judge's decision and the average WTP among those who stay. As illustrated in Appendix Figure A.10a, the WTP increases sharply with the employer quality.

7.2 The Value of Outside Options

Next, we examine how the value of the employer depends on whether the firm is liquidated in a strong or weak local market. In the data, there are 0.578 jobs per job seeker in strong labor markets, and 0.178 in weak labor markets, measured at the time of distress. In the second and third columns of Table 9, we report how the average WTP and corresponding NPV of disposable income losses differ for employees facing strong and weak local market conditions. The average WTP in weak market conditions is approximately 15 percent higher than the WTP in strong market conditions, corresponding to an elasticity of -0.21 with respect to changes in the number of job openings per job seeker. This differential is partly driven by changes in the share of workers who remain at the distressed firm at the time of the judge's decision: About 17 percent of employees avoid layoffs through employer-to-employer transitions in weak markets, whereas 37 percent avoid layoff in strong labor markets, lowering the average WTP. As expected, Appendix Figure A.10b shows that the share of workers leaving the employer prior to the judge's decision declines with job quality, except at the lowest rung, where workers have almost no value to lose.

The model also allows us to decompose the average WTP into leavers (with a WTP of zero) and workers who stay in troubled firms (i.e., the stayers). The last row of Table 9 reports the WTP among stayers, showing that selection on future layoff costs is particularly prevalent in strong labor markets.

³³This amount reflects risk aversion and added income uncertainty following employer shutdown as well as the disutility incurred when exerting search effort as workers seek to re-climb the ladder. We quantify the role of each in columns (1) and (2) of Panel (b) of Appendix Table A.13 by computing counterfactual WTPs, removing these forces one at a time. Column (1) of Appendix Table A.13 reports the WTP from an identical sample of workers but under the assumption that workers are risk neutral. Turning off the desire for consumption smoothing makes it easier for workers to recover from the effects of liquidation in utility terms, as they can more effectively offset current-period losses by increasing future consumption. However, the costs of recovering lost employment positions ensure that workers still have a substantial WTP, even without risk aversion. Column (2) reports the counterfactual WTP we obtain when calculating the WTP among an otherwise identical-to-the-baseline sample, assuming that search effort incurs zero cost. Search costs play a dual role in calculating the WTP, as they reduce the worker's utility both directly and by prolonging the time it takes for workers to recover their pre-liquidation income level and thus have a slightly larger contribution to the baseline WTP figure compared to risk aversion. Combined, these two factors are sufficient to explain approximately one-third of the workers' baseline willingness to pay to avoid liquidation.

7.3 The Value of Self-Insurance

Next, we calculate how learning about pending layoffs affects welfare by constructing a counterfactual experiment without advance information. In this experiment, we set the share of workers who learn about the pending layoff to zero. This configuration effectively removes workers' ability to respond to the petition before the judge's decision. Because it holds the composition of workers fixed, we refer to it as the *ideal experiment*, as it represents a natural benchmark to evaluating the benefit of worker reallocation. We report the findings of this experiment in Panel III of Table 9, and simulated effects on employment, wage income, disposable income, and liquid assets are reported in Appendix Table A.12.

Our headline result is that the average WTP to avoid liquidation increases by 14 percent under the ideal experiment. Comparing the last two rows of Table 9 highlights that approximately onequarter of workers leave the distressed firm before the decision in the baseline model—and thus have a WTP of zero—while all workers in the no-information scenario remain at the firm when the employer shuts down. Importantly, stayers in the baseline model have a higher WTP than stayers in the experiment, suggesting that welfare costs would be almost five percent higher when focusing on stayers compared to the average WTP for the full population of employees.

7.4 Implications for Measured Layoff Costs

Our final exercise quantifies how knowledge about financial risk shapes the composition of workers who experience layoff. In Panel (a) of Figure 8, we illustrate how earnings costs vary across three scenarios. The dashed line represents the ideal experiment, where all workers, irrespective of outside options experience layoff. The top line includes workers who leave in anticipation of layoff, where worker transitions result in smaller reductions in wage income and in some cases may even lead to wage gains as workers find higher quality jobs due to the increase in search effort. The bottom line includes only the earnings change for stayers, who have relatively poor outside options and are the workers with the most to lose. The shaded area from the top represent the bias from attenuation, while the shaded area from the bottom reflect composition bias. On average, the measured layoff cost is 20 percent higher than the fitted model, and about 10 percent higher than the " ideal experiment". The averages over the five-year period are reported in Panels (a) through (c) in Appendix Table A.12.

Two important restrictions are worth noting when evaluating these calculations. First, we have imposed no restrictions on tenure in the firm. This restriction is important, as human capital accumulates over the career and deteriorates during unemployment, thus longer-tenured workers face larger earnings losses and have comparatively weaker outside options than shorter-tenured workers on average. Hence, imposing longer tenure requirements, as is typically done in layoff research, would both reduce the anticipation effect—bringing our quasi-experimental estimate closer to the ideal experiment—and increase the composition bias—further increasing the difference be-

tween the layoff estimate and ideal experiment.³⁴ Second, we allow no more than three months of advance information. Allowing for additional periods of advance information could further attenuate, if not eliminate, the estimated effect of liquidation on wage income. This attenuation would be driven by the increased likelihood of finding a suitable offer given additional search opportunities and by the fact that those searching to avoid layoff could afford to have higher reservation wages due to the increased horizon for search.



Figure 8: Effects of Liquidation on Wage Income by Experiment Type

Notes: This figure illustrates how our estimate of the effect of liquidation on wage income in years 0 to 4 differs across our counterfactuals. Figure 8a reports average estimates while Figures 8b and 8c report estimates in the sub-samples of workers facing strong and weak local labor market conditions, respectively. The solid black line in each figure represents the estimate from our baseline fitted model (i.e., the judge experiment), while the dashed line represents the results under our ideal experiment where workers receive no advance information about the pending liquidation and the dash-dotted line represents the results of a layoff regression, constructed by conditioning on workers who remain at the firm at the time the bankruptcy judge's decision is made. The light-grey shaded area between our baseline and ideal experiment estimates represents the attenuation of the estimated effect of liquidation due to workers' anticipatory search behavior. The dark-grey shaded area between our ideal experiment and standard layoff estimates represents the amplification of the estimated effect due to adverse selection on layoff costs (i.e., the workers who remain at the firm are those with the poorest outside options).

Finally, we illustrate how the composition effect depends on workers' outside options. Figure **8b** shows that learning about risk plays a much larger role in attenuating the estimated effect of liquidation in strong local markets, when workers have a greater ability to respond to information. As a consequence of this heightened responsiveness, the compositional changes are much more prevalent in strong markets. Figure **8c** demonstrates that both attenuation and composition biases are much smaller in weak markets, because the relative lack of job opportunities reduces workers' ability to respond to learning about a future layoff. This brings both judge-design and layoff regression evidence closer to the *ideal* layoff regression without attenuation or composition effects.

³⁴Appendix Figure A.9 performs a selection correction in which we measure both the effects of employer shutdown and the WTP in sub-samples defined by the worker's human capital level and then re-weight our results to match the distribution of human capital in the sample of unemployed workers who do not experience distress rather than the distribution among those satisfying the potential tenure condition. We find that correcting for this selection on human capital causes our baseline WTP estimate to fall from NOK439,000 to NOK428,000 and generates slightly smaller estimates of the impact on wage and disposable income.

8 Conclusion

In this paper, we assessed the costs of employer distress to employees. We built a novel data set measuring income, wealth, job opportunities, and quasi-experimental variation in bankruptcy risk in Norway to address concerns of selection and omitted variables. The key to our empirical approach was that bankruptcy judges are randomly assigned and vary in how often they liquidate a firm. We found that employer shutdown causes a five-year average earnings decline of 24 percent. We also found that about a quarter of employees avoided layoff by transitioning to a new employer before the judge decided to liquidate the firm. Moreover, strong local labor markets and high-wage employers are associated with higher pre-bankruptcy employee reallocation and lower layoff costs, indicating that worker mobility is an important insurance mechanism. By comparison, the insurance from taxes and public transfers reduced the income loss by an equal amount.

To understand the welfare benefits of employee turnover, we estimated a structural model with savings, learning about bankruptcy risk, and on-the-job search. Using the model, we found that workers were willing to pay about the same as the net present value of the income loss to keep their current employment. Heterogeneity in the value of an employer and outside options were key drivers of (i) worker turnover in the data and (ii) willingness to pay to save an employer in the model. Our main finding is that shutting off learning about financial risk increased the welfare cost of liquidation by 14 percent, indicating that the search process provides workers with important insurance against risk. Taken together, our calculations indicate that employers disclosing financial information to employees can yield important benefits to employees.

References

- AARONSON, DANIEL, AGARWAL, SUMIT, HOTCHKISS, JULIE L., & KELLEY, TAYLOR. 2019. Job displacement and financial outcomes. *Economics Letters*, **177**, 18–21.
- ABOWD, JOHN M, KRAMARZ, FRANCIS, & MARGOLIS, DAVID N. 1999. High wage workers and high wage firms. *Econometrica*, **67**(2), 251–333.
- AGRAWAL, AJAY, GANS, JOSHUA S, & GOLDFARB, AVI. 2023. Do we want less automation? *Science*, **381**(6654), 155–158.
- AGRAWAL, ASHWINI K, & MATSA, DAVID A. 2013. Labor unemployment risk and corporate financing decisions. *Journal of Financial Economics*, **108**(2), 449–470.
- AIZER, ANNA, & JOSEPH J. DOYLE, JR. 2013. Juvenile Incarceration, Human Capital and Future Crime: Evidence from Randomly-Assigned Judges. June.
- ANDERSEN, ASGER LAU, JENSEN, AMALIE SOFIE, JOHANNESEN, NIELS, KREINER, CLAUS THUS-TRUP, LETH-PETERSEN, SØREN, & SHERIDAN, ADAM. 2023. How Do Households Respond to Job Loss? Lessons from Multiple High-Frequency Datasets. *American Economic Journal: Applied Economics*, 15(4), 1–29.

- ARAUJO, ALOISIO PESSOA DE, FERREIRA, RAFAEL, LAGARAS, SPYRIDON, MORAES, FLAVIO, PONTICELLI, JACOPO, & TSOUTSOURA, MARGARITA. 2020. The Labor Cost of Pro-Labor Bias in Bankruptcy. *Available at SSRN 3757117*.
- ATKINSON, ANTHONY B., RAINWATER, LEE, & SMEEDING, TIMOTHY M. 1995. Income Distributions in OECD Countries: Evidence from the Luxembourg Income Study. *OECD Publications and Information Center*.
- AUDOLY, RICHARD, DE PACE, FEDERICA, & FELLA, GIULIO. 2022. Job Ladder, Human Capital, and the Cost of Job Loss. *FRB of New York Staff Report*.
- AUTOR, DAVID, KOSTOL, ANDREAS, MOGSTAD, MAGNE, & SETZLER, BRADLEY. 2019. Disability benefits, consumption insurance, and household labor supply. *American Economic Review*, **109**(7).
- BAGHAI, RAMIN, SILVA, RUI, THELL, VIKTOR, & VIG, VIKRANT. 2020. Talent in Distressed Firms: Investigating the Labor Costs of Financial Distress.
- BAGHAI, RAMIN P, SILVA, RUI C, THELL, VIKTOR, & VIG, VIKRANT. 2021. Talent in distressed firms: Investigating the labor costs of financial distress. *The Journal of Finance*, **76**(6), 2907–2961.
- BERGER, DAVID W, HERKENHOFF, KYLE F, KOSTØL, ANDREAS R, & MONGEY, SIMON. 2023. *An Anatomy of Monopsony: Search Frictions, Amenities and Bargaining in Concentrated Markets*. Tech. rept.
- BERNSTEIN, SHAI, COLONNELLI, EMANUELE, & IVERSON, BENJAMIN. 2019a. Asset allocation in bankruptcy. *The Journal of Finance*, **74**(1), 5–53.
- BERNSTEIN, SHAI, COLONNELLI, EMANUELE, GIROUD, XAVIER, & IVERSON, BENJAMIN. 2019b. Bankruptcy spillovers. *Journal of Financial Economics*, **133**(3), 608–633.
- BERTHEAU, ANTOINE, ACABBI, EDOARDO MARIA, BARCELO, CRISTINA, GULYAS, ANDREAS, LOMBARDI, STEFANO, & SAGGIO, RAFFAELE. 2022. The unequal cost of job loss across countries. *American Economic Review: Insights.*
- BHULLER, MANUDEEP, DAHL, GORDON B, LØKEN, KATRINE V, & MOGSTAD, MAGNE. 2020. Incarceration, recidivism, and employment. *Journal of Political Economy*, **128**(4), 1269–1324.
- BIRINCI, SERDAR, PARK, YOUNGMIN, PUGH, THOMAS, & SEE, KURT. 2023. Uncovering the Differences Among Displaced Workers: Evidence from Canadian Job Separation Records. Tech. rept.
- BLUNDELL, R., PISTAFERRI, L., & PRESTON, I. 2008. Consumption Inequality and Partial Insurance. *The American Economic Review*, **98**(5), pp. 1887–1921.
- BOHN, A. 2000. *Domstolsloven, Kommentarutgave* [Law of Courts, Annotated Edition]. Vol. In Norwegian. Oslo: Universitetsopplaget.

- BONFIM, DIANA, & NOGUEIRA, GIL. 2021. Corporate Reorganization as Labor Insurance in Bankruptcy. *Available at SSRN 3761746*.
- BRAXTON, J CARTER, & TASKA, BLEDI. 2023. Technological change and the consequences of job loss. *American Economic Review*, **113**(2), 279–316.
- BRAXTON, J CARTER, HERKENHOFF, KYLE F, & PHILLIPS, GORDON M. 2020. *Can the unemployed borrow? implications for public insurance.* Tech. rept. National Bureau of Economic Research.
- BRIS, ARTURO, WELCH, IVO, & ZHU, NING. 2006. The costs of bankruptcy: Chapter 7 liquidation versus Chapter 11 reorganization. *The journal of finance*, **61**(3), 1253–1303.
- BROWN, JENNIFER, & MATSA, DAVID A. 2016. Boarding a sinking ship? An investigation of job applications to distressed firms. *The Journal of Finance*, **71**(2), 507–550.
- BROWNING, M., & LETH-PETERSEN, S. 2003. Imputing consumption from income and wealth information. *The Economic Journal*, **113**(488), F282–F301.
- BURDETT, KENNETH, CARRILLO-TUDELA, CARLOS, & COLES, MELVYN. 2020. The cost of job loss. *The Review of Economic Studies*, **87**(4), 1757–1798.
- CAGGESE, ANDREA, CUÑAT, VICENTE, & METZGER, DANIEL. 2019. Firing the wrong workers: Financing constraints and labor misallocation. *Journal of Financial Economics*, **133**(3), 589–607.
- CALDWELL, SYDNEE, & DANIELI, OREN. 2023. Outside Options in the Labor Market. *Review of Economic Studies*.
- CARRINGTON, WILLIAM J, & FALLICK, BRUCE. 2017. Why do earnings fall with job displacement? *Industrial Relations: A Journal of Economy and Society*, **56**(4), 688–722.
- CEDERLÖF, JONAS. 2020. Saved by Seniority? Effects of Displacement for Workers at the Margin of Layoff. *University of Uppsala Working Paper*.
- CEDERLÖF, JONAS, FREDRIKSSON, PETER, NEKOEI, ARASH, & SEIM, DAVID. 2021. Mandatory Advance Notice of Layoff: Evidence and Efficiency Considerations.
- CHE, YEON-KOO, & TERCIEUX, OLIVIER. 2021. Optimal Queue Design. Working Paper, 1, 45.
- CHETTY, RAJ, & SZEIDL, ADAM. 2007. Consumption commitments and risk preferences. *The Quarterly Journal of Economics*, **122**(2), 831–877.
- CHIAPPORI, PIERRE-ANDRE, & PAIELLA, MONICA. 2011. Relative Risk Aversion is Constant: Evidence From Panel Data. *Journal of the European Economic Association*, **9**(6), 1021–1052.
- CULLEN, J., & GRUBER, J. 2000. Does Unemployment Insurance Crowd out Spousal Labor Supply? *Journal of Labor Economics*, **18**(3), pp. 546–572.

- CUTLER, DAVID M, & KATZ, LAWRENCE F. 1992. Rising Inequality? Changes in the Distribution of Income and Consumption in the 1980's. *The American Economic Review*, **82**(2), 546.
- DAHL, G., KOSTØL, A., & MOGSTAD, M. 2014. Family Welfare Cultures. *The Quarterly Journal of Economics*, **129**(4), 1711–1752.
- DAVIS, STEVEN J, & VON WACHTER, TILL. 2011. Recessions and the Costs of Job Loss. *Brookings Papers on Economic Activity*, **2011**(2), 1–72.
- DE CHAISEMARTIN, CLEMENT. 2017. Tolerating defiance? Local average treatment effects without monotonicity. *Quantitative Economics*, **8**(2), 367–396.
- DOBBIE, WILL, & SONG, JAE. 2015. Debt relief and debtor outcomes: Measuring the effects of consumer bankruptcy protection. *American Economic Review*, **105**(3), 1272–1311.
- DOYLE, JOSEPH. 2007. Child Protection and Child Outcomes: Measuring the Effects of Foster Care. *The American Economic Review*, **97**(5), 1583–1610.
- EIKA, LASSE, MOGSTAD, MAGNE, & VESTAD, OLA L. 2020. What can we learn about household consumption expenditure from data on income and assets? *Journal of Public Economics*, **189**, 104163.
- FABERMAN, JASON R., MUELLER, ANDREAS I., SAHIN, AYSEGUL, & TOPA, GIORGIO. 2022. Job Search Behavior Among the Employed and Non-Employed. *Econometrica*, **90**(4), 1743–1779.
- FLAAEN, AARON, SHAPIRO, MATTHEW D, & SORKIN, ISAAC. 2019. Reconsidering the consequences of worker displacements: Firm versus worker perspective. *American Economic Journal: Macroeconomics*, **11**(2), 193–227.
- FRIEDRICH, BENJAMIN, LAUN, LISA, MEGHIR, COSTAS, & PISTAFERRI, LUIGI. 2019. *Earnings dynamics and firm-level shocks*. Tech. rept.
- GALASSO, ALBERTO, & SCHANKERMAN, MARK. 2015. Patents and cumulative innovation: Causal evidence from the courts. *The Quarterly Journal of Economics*, **130**(1), 317–369.
- GANONG, PETER, & NOEL, PASCAL. 2019. Consumer spending during unemployment: Positive and normative implications. *American Economic Review*, **109**(7), 2383–2424.
- GIBBONS, ROBERT, & KATZ, LAWRENCE F. 1991. Layoffs and lemons. *Journal of labor Economics*, **9**(4), 351–380.
- GRUBER, JONATHAN. 1997. The Consumption Smoothing Benefits of Unemployment Insurance. *The American Economic Review*, **87**(1), 192–205.
- HALLA, MARTIN, SCHMIEDER, JULIA, & WEBER, ANDREA. 2020. Job displacement, family dynamics, and spousal labor supply. *American Economic Journal: Applied Economics*, **12**(4), 253–87.

- HEATHCOTE, JONATHAN, STORESLETTEN, KJETIL, & VIOLANTE, GIOVANNI L. 2014. Consumption and labor supply with partial insurance: An analytical framework. *American Economic Review*, **104**(7), 2075–2126.
- HENDREN, NATHANIEL. 2017. Knowledge of future job loss and implications for unemployment insurance. *American Economic Review*, **107**(7), 1778–1823.
- HERKENHOFF, KYLE, PHILLIPS, GORDON, & COHEN-COLE, ETHAN. 2023. How credit constraints impact job finding rates, sorting & aggregate output. *Review of Economic Studies*.
- HUCKFELDT, CHRISTOPHER. 2022. Understanding the scarring effect of recessions. *American Economic Review*, **112**(4), 1273–1310.
- HUMPHRIES, JOHN ERIC, MADER, NICHOLAS S, TANNENBAUM, DANIEL I, & VAN DIJK, WIN-NIE L. 2019. *Does eviction cause poverty? Quasi-experimental evidence from Cook County, IL*. Tech. rept. National Bureau of Economic Research.
- IFERGANE, TOMER, et al. 2022. Time to Say Goodbye: The Macroeconomic Implications of Termination Notice. CFM, Centre for Macroeconomics.
- IMBENS, GUIDO W., & RUBIN, DONALD B. 1997. Estimating Outcome Distributions for Compliers in Instrumental Variables Models. *The Review of Economic Studies*, **64**(4), 555–574.
- JACOBSON, LOUIS S, LALONDE, ROBERT J, & SULLIVAN, DANIEL G. 1993. Earnings losses of displaced workers. *The American economic review*, 685–709.
- JAROSCH, GREGOR. 2022. Searching for job security and the consequences of job loss. Econometrica.
- KEYS, BENJAMIN J. 2018. The credit market consequences of job displacement. *Review of Economics and Statistics*, **100**(3), 405–415.
- KLING, JEFFREY R. 2006. Incarceration Length, Employment, and Earnings. *The American Economic Review*, **96**(3), 863–876.
- LACHOWSKA, MARTA, MAS, ALEXANDRE, & WOODBURY, STEPHEN A. 2020. Sources of displaced workers' long-term earnings losses. *American Economic Review*, **110**(10), 3231–66.
- LANDAIS, CAMILLE. 2015. Assessing the welfare effects of unemployment benefits using the regression kink design. *American Economic Journal: Economic Policy*, **7**(4), 243–78.
- LISE, JEREMY. 2013. On-the-job search and precautionary savings. *Review of Economic Studies*, **80**(3), 1086–1113.
- LIU, KAI. 2019. Wage risk and the value of job mobility in early employment careers. *Journal of Labor Economics*, **37**(1), 139–185.

- LJUNGQVIST, LARS, & SARGENT, THOMAS J. 1998. The European unemployment dilemma. *Journal of political Economy*, **106**(3), 514–550.
- LOW, HAMISH, MEGHIR, COSTAS, & PISTAFERRI, LUIGI. 2010. Wage risk and employment risk over the life cycle. *American Economic Review*, **100**(4), 1432–67.
- MATSA, DAVID A. 2018. Capital structure and a firm's workforce. *Annual Review of Financial Economics*, **10**, 387–412.
- MEGHIR, C., & PISTAFERRI, L. 2011. Earnings, Consumption and Life Cycle Choices. Handbook of Labor Economics, vol. 4, Part B. Elsevier.
- MOGSTAD, MAGNE, TORGOVITSKY, ALEXANDER, & WALTERS, CHRISTOPHER R. 2021. The causal interpretation of two-stage least squares with multiple instrumental variables. *American Economic Review*, Forthcoming.
- MUELLER, ANDREAS I, & SPINNEWIJN, JOHANNES. 2023. Expectations data, labor market, and job search. *Handbook of Economic Expectations*, 677–713.
- NEAL, DEREK. 1995. Industry-specific human capital: Evidence from displaced workers. *Journal* of labor Economics, 653–677.
- ROSE, EVAN K, & SHEM-TOV, YOTAM. 2023. How Replaceable Is a Low-Wage Job?
- RUHM, CHRISTOPHER J. 1992. Advance notice and postdisplacement joblessness. *Journal of Labor Economics*, **10**(1), 1–32.
- RUHM, CHRISTOPHER J. 1994. Advance notice, job search, and postdisplacement earnings. *Journal of Labor Economics*, **12**(1), 1–28.
- SAPORTA-EKSTEN, ITAY. 2014. Job loss, consumption and unemployment insurance. *Unpublished manuscript*, **1**, 50.
- SCHMIEDER, JOHANNES F, & VON WACHTER, TILL. 2016. The effects of unemployment insurance benefits: New evidence and interpretation. *Annual Review of Economics*, **8**, 547–581.
- SCHMIEDER, JOHANNES F, VON WACHTER, TILL M, & HEINING, JÖRG. 2023. The costs of job displacement over the business cycle and its sources: evidence from Germany. *American Economic Review*, **113**(5).
- SIMMONS, MICHAEL STEWART. 2021. Pre-Layoff Search.
- STEPHENS, MELVIN JR. 2004. Job loss expectations, realizations, and household consumption behavior. *Review of Economics and statistics*, **86**(1), 253–269.
- STEPNER, MICHAEL. 2019. The insurance value of redistributive taxes and transfers. *Unpublished manuscript*.

For Online Publication:

Online Appendix:

Layoff Costs and Learning About Employers' Financial Distress

BY ANDREAS R. KOSTØL, MATTHEW MERKLE, AND MORTEN GRINDAKER,

A Appendix: Additional Tables and Figures

	(I) Bas	seline instrument	(II) Reve	erse-sample instrument		
Dependent variable		Pr(Allow)	Pr(Allow)			
Young workers (below 35)	.948	Dep. mean: .538	.890	Dep. mean: .539		
	(.026)	N: 24796	(.032)	N: 22761		
Older workers	.926	Dep. mean: .582	.847	Dep. mean: .584		
	(.023)	N: 29221	(.029)	N: 26043		
Females	.935	Dep. mean: .572	.871	Dep. mean: .573		
	(.040)	N: 13053	(.046)	N: 12215		
Males	.906	Dep. mean: .559	.685	Dep. mean: .557		
	(.028)	N: 40964	(.043)	N: 29991		
Married	.932	Dep. mean: .581	.877	Dep. mean: .584		
	(.023)	N: 22836	(.027)	N: 21092		
Unmarried	.934	Dep. mean: .548	.864	Dep. mean: .552		
	(.026)	N: 31181	(.032)	N: 27666		
Foreign-born	.896	Dep. mean: .537	.774	Dep. mean: .542		
C	(.035)	N: 9678	(.043)	N: 9137		
Native-born	.926	Dep. mean: .567	.706	Dep. mean: .557		
	(.035)	N: 44339	(.054)	N: 29884		
Less than high school	.917	Dep. mean: .552	.872	Dep. mean: .555		
0	(.027)	N: 23167	(.031)	N: 21287		
At least high school	.948	Dep. mean: .570	.897	Dep. mean: .574		
0	(.024)	N: 30850	(.030)	N: 27914		
High wage	.959	Dep. mean: .549	.872	Dep. mean: .553		
0 0	(.024)	N: 27007	(.037)	N: 25021		
Low wage	.883	Dep. mean: .575	.767	Dep. mean: .573		
0	(.031)	N: 27010	(.040)	N: 23384		
Many Employees	1.27	Dep. mean: .587	.468	Dep. mean: .576		
5 I 5	(.053)	N: 26438	(.160)	N: 23370		
Few Employees	.434	Dep. mean: .537	.101	Dep. mean: .537		
I I I I I I I I I I	(.060)	N: 27579	(.044)	N: 14481		

Table A.1: Cross-Sample First Stage Estimates

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Notes: This table reports heterogeneity in first stage estimates using the baseline instrument (I) and the reverse-sample instrument (II). The first stage specification in (I) corresponds to subsample estimates of the first stage using our baseline judge stringency measure. The reverse-sample instrument (II) is constructed by calculating judge stringency using all cases *except* for those in the specified subsample (e.g., judge stringency for the subsample of older applicants is constructed using judges' decisions for younger applicants). The baseline and the reverse sample instrument in this exercise are based on all firm cases with employees (including cases outside of our estimation sample but not including firms without employees).

Years after decision (<i>t</i>):	0	1	2	3	4		
Panel (a):	Full Time Work						
Judge stringency	-0.426***	-0.254***	-0.104	-0.062	-0.058		
	(0.112)	(0.082)	(0.069)	(0.067)	(0.061)		
Overall mean of dependent variable	0.675	0.662	0.670	0.673	0.672		
Panel (b):	Any Employment						
Judge stringency	-0.456***	-0.223***	-0.132**	-0.104*	-0.078		
	(0.105)	(0.076)	(0.066)	(0.059)	(0.060)		
Overall mean of dependent variable	0.704	0.708	0.723	0.733	0.731		
Panel (c):		Total Wage	Income (N	OK1,000)			
Judge stringency	-100.082**	-97.198**	-83.039**	-88.340**	-56.444		
	(39.578)	(48.547)	(41.461)	(42.036)	(40.507)		
Overall mean of dependent variable	429.722	403.118	402.357	406.839	407.947		
Number of observations:	14,826	14,826	14,826	14,826	14,826		

Table A.2: Reduced Form Effects on Employment

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the reduced form estimates of judge stringency on employment and total wage income. The sample includes firms with at least five employees before the petition is received and full-time employees aged 25–62. There are 312 judges. Each regression controls for fully interacted year and court indicators, a linear and quadratic of leave-out judge processing time, and individual control variables (see Table 4 for details). Employment is defined as having any wage income in a year. Unemployment is receiving at least NOK1,000 in unemployment benefits over the calendar year.

Table A.3: 2SLS Estimates: Worker Mobility Rates

Years after decision (<i>t</i>): 4			
	(1)	(2)	(3)
Dependent variable:	Occupational	Industry	Geographical
	Mobility	Mobility	Mobility
Liquidation (β)	-0.030	-0.024	0.003
	(0.132)	(0.045)	(0.048)
Overall mean of dependent variable	0.565	0.370	0.081
Untreated mean	0.520	0.449	0.042
Number of observations	14,826	14,826	14,826

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. The sample includes firms with at least five employees the year before the petition. The estimation sample includes full-time employees aged 25–62 employed 12 months prior to petition. There are 312 judges. Each regression controls for fully interacted year and court indicators, a linear and quadratic of leave-out judge processing time, and individual control variables (see Table 4 for details).

Years after decision (<i>t</i>): 4				
	(1)	(2)	(3)	(4)
Dependent variable:	UI	DI	Any Labor	Labor Force
	Receipt	Receipt	Earnings	Participation
Liquidation (β)	0.025	-0.025	-0.103*	-0.107*
	(0.056)	(0.024)	(0.058)	(0.058)
Overall mean of dependent variable	0.117	0.027	0.868	0.876
Untreated mean	0.107	0.013	0.937	0.951
Number of obs. 15,073	14,826	14,826	14,826	14,826

Table A.4: 2SLS Estimate: Labor Force Participation

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the second stage coefficients of equation 2, overall, and untreated means of the dependent variable four years after the bankruptcy petition. The sample includes firms with at least five employees before the petition is received and full-time employees aged 25-62. There are 312 judges. Each regression controls for fully interacted year and court indicators, a linear and quadratic of leave-out judge processing time, and individual control variables (see Table 4 for details). Full-time employment is equal to one if the year-end work contract states at least 30 hours per week and is otherwise zero. Disability insurance receipt is an indicator equal to 1 if a person receives any DI benefits during the year. The indicator for any labor earnings equals 1 if the person has at least NOK1 in labor earnings in a given year. Labor force participation equals having any labor earnings or receiving any unemployment benefits in a given year.

Dependent variable:	A. Occupation Move	B. Industry Move	C. Geography Move
Liquidation (β)	-0.030	-0.024	0.003
	(0.132)	(0.045)	(0.048)

0.370

0.449

14,826

0.565

0.520

14,826

0.081

0.042

14,826

Untreated mean

Number of observations

Overall mean of dependent variable

Table A.5: Sectoral Mobility

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the 2SLS estimates of equation 2 and the average of the dependent variable for the overall and untreated samples. The sample includes firms with at least five employees the year before the petition. The estimation sample includes full-time employees aged 25-62 employed 12 months prior to petition. There are 312 judges. Each regression controls for fully interacted year and court indicators, a linear and quadratic of leave-out judge processing time and individual control variables (see Table 4 for details). Mobility is measured as the cumulative chance of being observed in a different occupation, industry, or commuting zone over the four years after the petition.

	A. Occu	pation FE and	Individual	Controls		B. Occupa	ation FE	
	I. Weal	k Market	II. Stron	g Market	I. Wea	k Market	II. Stron	g Market
	Firm	Wage	Firm	Wage	Firm	Wage	Firm	Wage
	Closes	Income	Closes	Income	Closes	Income	Closes	Income
	(1)	(3)	(1)	(3)	(1)	(3)	(1)	(3)
Judge Stringency	0.930***	-122.783**	0.743***	-46.593	0.923***	-118.096**	0.721***	-48.375
	(0.250)	(54.939)	(0.227)	(44.011)	(0.250)	(53.769)	(0.224)	(45.046)
Liquidation (β)		-131.981**		-62.692		-127.951**		-67.096
		(57.738)		(59.458)		(56.592)		(63.424)
Overall mean	0.546	385.255	0.559	431.102	0.549	385.058	0.556	431.218
Untreated mean		461.875		448.321		461.679		454.686

Table A.6: Alternative Methods to Residualize Vacancy-Unemployment Ratios

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the first stage, reduced form, and second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. Untreated means are potential outcome of employees in marginal firms (see Dahl *et al.*, 2014). The sample includes firms with at least five employees the year before the petition. The estimation sample includes full-time employees aged 25–62 employed 12 months prior to petition. There are 312 judges. Each regression controls for fully interacted year and court indicators and includes the individual demographic control variables from Table 2 and dummies for each quintile of prepetition wages. Average exchange rate in 2024 is NOK/\$=10.

Table A.7: 2SLS Estimates: H	eterogeneity
------------------------------	--------------

	A. Abo	ve Age 40	B. Belov	B. Below Age 40		C. Low Skilled		D. High Skilled	
	Firm	Wage	Firm	Wage	Firm	Wage	Firm	Wage	
	Closes	Income	Closes	Income	Closes	Income	Closes	Income	
	(1)	(3)	(1)	(3)	(1)	(3)	(1)	(3)	
Judge Stringency	0.591***	-136.210**	0.966***	-47.718	0.737***	-69.926*	0.811***	-83.002*	
	(0.224)	(55.525)	(0.210)	(36.606)	(0.219)	(42.106)	(0.215)	(43.781)	
Liquidation (β)		-230.536**		-49.375		-94.928		-102.363**	
		(110.542)		(36.852)		(58.357)		(51.439)	
Overall mean	0.593	420.667	0.519	400.889	0.549	322.655	0.555	462.021	
Untreated mean		529.514		424.839		366.773		493.737	
Pre-layoff turnover	0	.247	0.2	250	0.2	233	0.	203	
Observations:	6	,817	7,9	992	5,5	526	9,	281	

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the first stage, reduced form, and second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. Untreated means are potential outcome of employees in marginal firms (see Dahl *et al.*, 2014). The sample includes firms with at least five employees the year before the petition. The estimation sample includes full-time employees aged 25–62 employed 12 months prior to petition. There are 312 judges. Each regression controls for fully interacted year and court indicators and includes the individual demographic control variables from Table 2 and dummies for each quintile of prepetition wages. The mean wage income is winsorized at the 99th percentile and disposable income is winsorized at the 95th percentile. Outcomes are averaged over the five-year period and are deflated to 2015-values using consumer price index. Median age is 40 years. Low skilled is having less than high-school and high skill is high school or more. Average exchange rate in 2024 is NOK/\$=10.

Years after decision (<i>t</i>):	0	1	2	3	4
Liquidation (β)	0.046	0.061	0.067	0.022	-0.028
	(0.056)	(0.059)	(0.061)	(0.065)	(0.075)
Overall mean of dependent variable	0.378	0.389	0.400	0.408	0.414
Untreated mean	0.339	0.346	0.359	0.394	0.438
Number of observations:	14,826	14,826	14,826	14,826	14,826

Table A.8: 2SLS Estimates: Currently Married

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. The sample includes firms with at least five employees the year before the petition. The estimation sample includes full-time employees aged 25–62 employed 12 months prior to petition. There are 312 judges. Each regression controls for fully interacted year and court indicators, a linear and quadratic of leave-out judge processing time, and individual control variables (see Table 4 for details). Marital status is measured by the end of the year and does not account for co-habitation.

Years after decision (<i>t</i>):		4		0–4	0–4	0–4
	(1)	(2)	(3)	(4)	(5)	(6)
Dependent variable:	Δ	Δ	Δ	Business	Capital	Stock
	Net Assets	Active	Debt	Income	Income	Income
Liquidation (β)	-296.550**	-207.374*	-46.448	-0.191	-0.886	-4.660
	(147.991)	(115.727)	(138.873)	(0.954)	(0.980)	(14.183)
Overall mean of dependent variable	498.733	332.833	316.679	1.406	3.495	32.777
Untreated mean	557.069	324.391	448.758	2.018	3.266	17.991
Number of obs.	14,826	14,826	14,826	14,826	14,826	14,826
15,073						

Table A.9: Impacts on Components of Imputed Consumption

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. The sample includes firms with at least five employees the year before the petition. The estimation sample includes full-time employees aged 25–62 employed 12 months prior to petition. There are 312 judges. Each regression controls for fully interacted year and court indicators, a linear and quadratic of leave-out judge processing time, and individual control variables (see Table 4 for details). Outcome variables are deflated to 2015-values using consumer price index and winsorized at 95th percentile. Average exchange rate in 2024 is NOK/\$=10.

Years after decision (<i>t</i>): 4					
	Δ	Net	Personal	Wage	Average
	Debt	Debtor	Bankruptcy	Garnishment	Interest Rates
	(1)	(2)	(3)	(4)	(5)
Liquidation (β)	-46.448	0.080	0.007	0.046	-0.007
	(138.873)	(0.077)	(0.011)	(0.059)	(0.008)
Overall mean of dependent variable	316.679	0.339	0.004	0.118	0.046
Untreated mean	448.758	0.313	-0.006	0.148	0.039
Number of observations:	14,826	14,826	14,826	14,826	14,826

Table A.10: 2SLS Estimates: Credit Standing

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: This table reports the second stage coefficients of equation 2 and the average of the dependent variable for the overall and untreated samples. The sample includes firms with at least five employees the year before the petition. The estimation sample includes full-time employees aged 25–62 employed 12 months prior to petition. There are 312 judges. Each regression controls for fully interacted year and court indicators, a linear and quadratic of leave-out judge processing time, and individual control variables (see Table 4 for details). Personal bankruptcy is equal to one if the person is granted debt relief, and otherwise zero. Net debtor is equal to one if debt exceeds total wealth, and otherwise zero. Wage garnishment is equal to one if the person is observed with any garnishment during the year, and otherwise zero. Car collateral is an indicator variable for whether someone uses their car as collateral for a loan. All variables are deflated to 2015-values using consumer price index. Interest rates are calculated from top 1% coded interest payments over total year-end debt. Average exchange rate in 2024 is NOK/\$=10.

Panel (a):	Wage Income (Average 0–4 NOK1,000)								
	(1)	(2)	(3)	(4)	(5)	(6)			
Liquidation (β)	-107.014**	-113.079**	-109.235**	-102.762***	-106.490*	-121.633*			
	(48.702)	(50.746)	(49.229)	(32.393)	(55.444)	(66.674)			
Panel (b):		Disposab	le Income (Av	verage 0–4 NC	DK1,000)				
	(1)	(2)	(3)	(4)	(5)	(6)			
Liquidation (β)	-52.557**	-55.315**	-58.752**	-39.677***	-55.614**	-87.167			
	(22.859)	(26.058)	(23.300)	(14.501)	(25.736)	(54.101)			
F-value	15.45	9.84	7.92	10.73	6.436	15.45			
Court*Year FE	Yes	Yes	Yes	Yes	Yes	Yes			
Individual Controls	Yes	Yes	Yes	Yes	Yes	Yes			
Minimum Cases per Judge	20	30	20	20	20	20			
First Stage Specification	Linear	Linear	Quadratic	Cubic	Linear	Linear			
Judge Processing time	Quadratic	Quadratic	Quadratic	Quadratic	Cubic	Quadratic			
Winsorizing	Yes	Yes	Yes	Yes	Yes	No			
Observations	14,826	12,002	14,826	14,826	14,826	14,826			

Table A.11: Robustness of 2SLS Estimates

***p<.01, **p<.05, *p<.10. Standard errors (in parentheses) are clustered at the judge-year level.

Note: The sample includes firms with at least five employees the year before the petition. The estimation sample includes full-time employees aged 25–62 employed 12 months prior to petition. There are 312 judges. Regressions always control for year and court indicators and dummies for each age, educational group, marital, immigration status, pre-petition wage quintile, and a quadratic function of leave-out judge processing time, unless otherwise specified. Wage income is winsorized at 99th and disposable income is winsorized 95 percentile levels and deflated to 2015-values using consumer price index. Average exchange rate in 2024 is NOK/\$=10.

Outcome:	Employment		Wage Income		Disposable Income		Liquid Assets	
	Level	Share	Level	Share	Level	Share	Level	Share
(a) Fitted Model	-0.113	-0.122	-179.627	-0.247	-75.384	-0.154	-76.552	-0.568
(b) Ideal Experiment	-0.130	-0.141	-201.880	-0.276	-84.308	-0.171	-83.838	-0.651
(c) Mass Layoff	-0.146	-0.156	-220.450	-0.295	-90.983	-0.182	-95.813	-0.652

Table A.12: Average Effects of Liquidation by Market and Experiment Type

Years after decision (*t*): average 0–4

Notes: This table shows the estimated effect of treatment on outcomes in year 0 through year 4 as a level and as a percentage of the untreated average value. We report results from our baseline fitted model, from the counterfactual model where workers do not receive any advance notice that their firm is in distress (i.e., the ideal experiment), and in our baseline fitted model conditioning on workers remaining at the firm at the of the judge's decision (i.e., a mass-layoff). Average exchange rate in 2024 is NOK/\$=10.

	Panel (a):	Panel (b):		
		No Risk	No Search	
	Baseline	Aversion	Costs	Residual
		(1)	(2)	(3)
Willingness to Pay	439.740	394.756	343.862	
Difference		44.985	95.878	298.877
As Share of Baseline		10.230%	21.803%	67.967%

Notes: This table reports the value of keeping the current employment position to workers in distressed firms. Panel (a) reports the average amount of liquid assets that the worker would be willing to forgo to have their firm's bankruptcy case dismissed by the judge rather than undergoing liquidation. Panel (b) decomposes the sources of workers' WTP using counterfactual experiments. Columns (1) and (2) report the estimated WTP under counterfactuals where workers are assumed to be risk-neutral and to face zero search costs, respectively, and report the shares of our baseline WTP figure which can be attributed to these two components. Column (3) reports the residual, which includes any potential interaction effects between risk aversion and search costs (which may be either positive or negative) as well as other factors such as the persistent effects of liquidation on wage income.





Notes: The figures plots unemployment rate for the adult population from the Labor Force Survey in Norway from 1992 to 2021. The series are from Statistics Norway.



Figure A.2: Firm Size Distribution

Notes: This figure plots the number of full-time employees aged 25-62 in firms petitioned for bankruptcy in 2006-2013.





Notes: Panel A.3a shows the average change in sales relative to the petition date for our baseline sample of firms and Panel A.3b shows the distribution of annual sales growth and the relationship between petition rates and sales.





Notes: Baseline specification removes controls for fixed effects for gender, commuting zone and occupations, and a polynomial of age and tenure when obtaining the (residualized) vacancy-unemployment ratio. Alternative specification 1 controls for gender and occupation fixed effects, and a polynomial of age and tenure. Alternative specification 2 only controls for occupation fixed effects.



Figure A.5: Heterogeneous Wage Impacts by Pre-Petition Unemployment Risk

Notes: This figure plots heterogeneity in impacts of shutting down a firm on wage income against the fraction of workers who experience unemployment at some point from 12 months before up to month of the petition. The size of scatter plots reflect the inverse of the standard errors of the estimated impact on disposable income.



Figure A.6: Impacts of Layoff on Income

Notes: This figure plots IV estimates on sources of income, all normalized to percent of pre-layoff level of gross earnings. Average exchange rate in 2024 is NOK/USD \approx 10.





Notes: These figures show the observed and offered wage distributions for each skill type. Observed wage uses residualized wages and the empirical distribution of employees at each wage level. Offered wage uses residualized vacancy-unemployment rates within 3-digit occupation and commuting zone cells as weights. Residualized wages and vacancy-unemployment ratios are normalized to the average and controls for gender, commuting zone- and occupation-fixed effects, and a quadratic function of age and tenure. Average exchange rate in 2024 is NOK/\$=10.





Notes: These two figures show the simulated effect of layoff on disposable income and liquid assets as a share of pre-layoff average annual wage income.





Notes: These figures plot the estimated effects of employer shutdown on both wage income and disposable income in our baseline fitted model and when correcting for selection on human capital due to the restrictions on our simulated bankruptcy sample. The baseline fitted model conditions the bankruptcy sample on workers having at least 12 months of potential tenure at the firm at the time the judge's decision is made. To construct the selection corrected effects, we estimate the effects on sub-samples corresponding to each level of human capital in the model and then take a weighted-average of these sub-sample estimates based on the distribution of human capital among workers experiencing the standard exogenous separation shock. Effects are reported as a fraction of the untreated complier mean.



Figure A.10: Heterogeneity

Notes: These figures plot how the WTP and share of workers leaving the distressed firm prior to the judge's decision varies with the quality of the employer (i.e., the rung of the wage ladder).

B Data Sources

B.1 Data Sources

Judicial System/DA. This data set provides us with information on cases related to bankruptcy. It provides us with the legal paragraphs, an anonymized identifier for the assigned judge, dates in which the case was received, and decided. It includes some information on creditors, and it includes some information on the proceeds of the court. The leave-out mean judge instrument is based on the full set of cases which a judge has handled.

Administrative Population Registers. Information on workers' geographic location, education, annual earnings, and household income. Social security data is provided by longitudinal administrative registers provided by Statistics Norway. This data covers every Norwegian resident from 1967 to 2019, and contain individual demographic information (including sex, age, zip codes, and years of education) and, since 1993, all sources of annual income, including earnings, self employment income, capital income, taxes, and cash transfers. Most of the income is third-party reported to tax authorities.

Administrative Employer-Employee Register. Workers' earnings and employment histories, and transitions between jobs come from the Norwegian Matched Employer-Employee Register maintained by Statistics Norway. The data is reported by the employer to tax authorities at the end of the year. The reporting was automated at the firm level and based on monthly payments after 2015 and covers a slightly broader set of low-paying jobs. Before 2015 contracts with fewer than 4 hours per week or below an annual amount of NOK10,000 were not reported. The data includes information on the dates of alterations to the contract, and the corresponding wage, industry and occupational codes, geographic location, and tenure at the establishment.

B.2 Sample Construction

There are 102 961 unique bankruptcy cases and 1 652 different judges. We exclude cases where the firm is involved in several cases in a given year and keep the bankruptcy cases where we observe either a manager or a managing owner. We also exclude cases where a manager appears in several cases each year and only keep the first observation for each firm or manager. After these restrictions, there are 60,822 bankruptcy petitions from which the judge stringency measure is generated.

B.3 Estimating Jobs at Risk of Bankruptcy

To estimate the US series, we collect the Quarterly Bankruptcy Filings from US courts for each year from 2001 to 2023. We collect the total US Chapter 7 and Chapter 11 filings where the nature of debt is business if the debtor is a corporation or partnership, or if debt related to the operation

of a business predominates. To get the jobs at risk, we first divide the total number of filings by the yearly average of the monthly level of employment from the CPS. Our estimates are obtained by further multiplying the annual Chapter 7 rate per employee by 72, the average number of employees in Chapter 7 firms, and 309 for Chapter 11 firms, as reported by Bernstein *et al.* (2019b).

To compute the numbers from Norway, we measure the total employment directly before the petition is received. Figure B.1 illustrates a similar pattern for Norway as in the US. The fraction of jobs at risk in bankruptcy but dismissed by a judge peaked at almost 15 percent percent during the Great Recession. In contrast, the jobs at risk for liquidation show a slower increase, peaking a few years later.





Notes: The figure plots the total number of employees exposed to either a liquidation or petition that is subsequently dismissed by a judge divided by the total workforce from 2005 to 2018.

C Mass Layoff Design

This section examines the earnings losses associated with a mass layoff event. We describe the data and sample construction and present the main findings.

C.1 Data and Sample Construction

To construct the mass layoff event, we proceed in two steps. In the first step, we compute a monthly dataset with firm employment using the matched employer-employee dataset from 2004 to 2014. We restrict the sample to full-time employees and only keep workers with one employment relationship in a given month. A mass layoff event follows from the definition in Lachowska *et al.* (2020). An employer is classified as having a mass layoff in a month if (i) employment dropped by 30 percent or more compared to the previous year's highest employment level, (ii) maximum employment in the previous year did not exceed 130 percent of the employment the year before, and (iii) employment at the month before the layoff exceeded 50. By this definition, we define 1234 firms that experienced a mass layoff during the sample period. Finally, we define the first month as the ML-event time.

In the second step, we identify all workers in these firms from 2000–2014. This dataset is then combined with information on their wage earnings and disposable income from the tax registry data and their age and education from Statistic Norway. Wage earnings and disposable income are deflated by CPI, with 2015 as the base year. To define the mass layoff sample of workers, we again follow Lachowska *et al.* (2020). A worker is defined as displaced if the worker is separated from his employer within 12 months of the lay-off event. The control group consists of workers who remained in the firm for at least 48 months after the event. As in Lachowska *et al.* (2020), the main sample consist of workers with at least six years of tenure in the firm.

Table C.1 reports the summary statistics for the employees of firms that experience a mass layoff event. Panel (a) reports averages and standard deviations for characteristics of the displaced workers and Panel (b) for the non-displaced workers. Compared to the workers in financially distressed firms in Table 1, workers in the mass layoff sample are older, somewhat lower educated, and higher paid. The sample is mainly concentrated in manufacturing industries and information, financial, and professional services.¹ Within the mass layoff sample, displaced workers have somewhat lower tenure and wage earnings than non-displaced workers.

¹This industry category consist of workers in information services (ITC), finance and insurance, professional and scientific services, and real estate.

Sample:	(a) Displaced			(b) Non-Displaced		
	Mean	St. Dev	Mean	St. Dev		
	(1)	(2)	(3)	(4)		
Age	48.04	(10.404)	47.16	(8.618)		
Female	0.35	(0.475)	0.32	(0.467)		
High School Graduate	0.39	(0.487)	0.40	(0.489)		
Bachelor Degree	0.19	(0.395)	0.18	(0.386)		
Higher Degree	0.08	(0.264)	0.09	(0.284)		
Tenure (years)	11.84	(6.557)	12.70	(6.585)		
Wage Income (NOK1,000)	541.49	(397.381)	584.34	(340.951)		
Industry (proportions)						
Mining, utilities, construction	0.074	(0.262)	0.093	(0.290)		
Manufacturing	0.327	(0.469)	0.357	(0.479)		
Trade and transportation	0.148	(0.355)	0.163	(0.369)		
Information, finance, prof. services	0.216	(0.412)	0.151	(0.358)		
Educational and health services	0.092	(0.288)	0.063	(0.243)		
Arts, recreation, hospitality services	0.056	(0.230)	0.067	(0.251)		
Number of workers	24,473		13,117			

Table C.1: Summary Statistics

Notes: This table reports summary statistics of our sample of workers in firms experiencing a mass layoff event and having at least six years of tenure. Tenure and wage earnings are measured the year prior to the mass layoff event. Panel (a) reports averages and standard deviations of key characteristics for displaced workers in these firms (separated from the firm within 12 months of the event). Panel (b) reports statistics for non-displaced workers (who remained in the firm for at least 48 months after the event). Workers that leave the firm between 12 and 48 months after the event are not included in the groups of displaced or non-displaced workers. Information, finance, and professional services include finance, insurance, real estate, professional, scientific services, and management of companies. Nominal values are deflated to 2015-values using the consumer price index. NOK/\$ \approx 9.

C.2 Event Study Estimates from Mass Layoff Sample

We begin by presenting the evolution of employment and wage earnings among the displaced and non-displaced workers in the sample. Figure C.1a plots the average employment for displaced and non-displaced workers five years prior to the mass layoff event and four years after. Four years after, only 85 percent of the treatment group (displaced workers) are employed. In Figure C.1b we make a similar plot of displaced and non-displaced workers' average wage earnings. There is a drop in wage earnings of approximately 50,000 NOK for displaced workers after the event, while the wage earnings of the non-displaced workers continue on an upward trajectory.

To estimate the earnings losses associated with the mass layoff event, we use the multi-period difference-in-difference estimator of Jacobson *et al.* (1993). This estimator compares the outcomes of displaced workers before, during, and after displacement with the control group of non-displaced workers. This estimator allows us to compare our estimates against the estimated earnings losses from the mass layoff literature. Following Jacobson *et al.* (1993) and Lachowska *et al.* (2020) we include worker and year fixed effects in our regression and estimate the following model:

$$Y_{it} = c_i + \gamma_t + \sum_{k=-5}^{4} (\delta_k * D_{itk}) + e_{it}$$
(8)

where Y_{it} is an employment outcome (employment or earnings) of worker *i* in year *t*, c_i is a workerspecific fixed effect, and γ_t is a vector of calendar year indicators. Each D_{itk} is an indicator equal to one when the worker is observed in year *t* relative to displacement. The coefficient of interest δ_k measures the difference in outcomes between displaced and non-displaced workers for each time *k* relative to the event. Figure C.2 displays the estimated employment and earnings difference (δ_k) for each year *k* relative to the mass layoff event. The event study estimates show an employment decline of approximately 15 percentage points four years after the event (C.3a) and a decline in wage earnings of approximately 150,000 NOK (C.3b). This constitutes an earnings loss of about 30 percent relative to the displaced workers' earnings the year prior to the event. This percentage decline is similar to the long term earnings losses reported in other studies (e.g, see Table 3 in Lachowska *et al.* (2020)).²

Figure C.1: Average Employment and Wage Earnings of Displaced and Non-Displaced Workers



Notes: This figure displays the average employment and wage earnings for displaced (treatment group) and non-displaced (control group) workers for the years t = -5 to t = 4 relative to the mass layoff event. Panel C.1a shows the evolution in employment and Panel C.1b shows the evolution in wage income.

In the standard mass layoff design, the control group is restricted to workers who remain in the firm throughout the estimation period. By construction, this restriction does not take into account that workers in the control group can lose their job or quit the firm after the event. To illustrate how this restriction affects the employment and wage estimates, Figure C.3 displays the estimated employment and earning effect in an alternative specification where we only restrict

²We also estimate the average wage and employment effect using the same specification as above, but using a dummy for the full post treatment period. The average estimated wage loss is 104.047 (2.278) NOK and the average estimated employment effect is 0.078 (0.001) percentage points.

non-displaced workers to remain in the firm for at least two years after the event. While the estimated employment effect remains fairly similar, the estimated decline in wage earnings is considerably lower with the alternative specification. Figure C.4b shows an estimated earnings decline of approximately 110,000 NOK. This is 25 percent lower than the earnings effect obtained by the standard specification. By ignoring that non-displaced workers might leave or lose their job after the mass-layoff, the estimated earnings effect in the standard specification appears to overstate the earnings losses of displaced workers.



Figure C.2: Control group: Workers who remain at least four years after event

Notes: This figure displays the estimated employment and earnings effect (δ_k) using equation 8. Panel C.3a displays the estimated effect on employment and Panel C.3b displays the estimated effect on wage earnings.





(a) Employment (b) Wage Income Notes: This figure displays the estimated employment and earnings effect (δ_k) using equation 8. Panel C.4a displays the estimated effect on wage earnings.

D Model Details

This section outlines additional details of our structural model. We present the full worker's problem including the worker's value function in each state and relevant constraints. We then provide the detailed results of our estimation of the model, including an outline of the 13 moments that are used in our Method of Simulated Moments estimation.

D.1 Worker's Value Function

We let *t* denote the age of the worker in months, which ranges from 1 to 504, and let V_t^i denote the worker's value at age *t* with employment status *i*. At each age, the worker observers her state and chooses an optimal level of net assets a^* , consumption c^* , and search effort e^* in order to maximize the expected discounted value of her utility, subject to a standard budget constraint and to the exogenous stochastic processes that determine the evolution of human capital, search outcomes, and the evolution of labor market conditions.

We let *a* denote the worker's level of net liquid-assets, which take the form of a one period bond with return normalized to one and price q_b when a < 0 and q_s when $a \ge 0$. We impose a borrowing limit <u>a</u> at the time of retirement, which ensures that workers do not borrow more than they can repay based on pension income.¹ In every preceding period, workers face a natural borrowing limit implied by the retirement borrowing constraint and non-negativity constraint on consumption. In addition to income from previously held bonds, workers collect income from the labor market in the form of a wage, an unemployment insurance benefit, or a basic social assistance benefit. The income of currently employed workers is determined by state variables wand h denoting the pre-tax base wage and human capital level of the worker, respectively. The income of currently unemployed workers is determine the level of UI benefit that the worker is eligible for—and the worker's choice of search effort e, which determines if the worker receives a UI benefit b(w, h) or a basic social assistance benefit <u>b</u> in that period.

Workers have a distress status *d* which describes the number of periods in which they have been aware that their firm is in distress. We let *d* take on value zero when the worker's firm is not involved in bankruptcy proceedings. Unemployed workers always have distress status d = 0by construction, while employed workers have distress status d = 0 when the firm is not at risk of bankruptcy or when the worker has not become aware of the risk. Once the worker becomes aware of distress, *d* denotes the number of periods that the worker has been aware of distress, which determines the worker's belief about when the judge's decision will occur. Workers with any non-zero value of *d* return to the non-distressed d = 0 status any time they leave the current employer through search or through the exogenous separation shock $\delta_{k,M}$. Workers transition from the non-distressed state to distress d = 1 with probability $\pi_{k,M}$ and transition from d > 0 to d = 0

¹Specifically, we set the borrowing limit <u>a</u> equal to the net present value of the lowest retirement pension income available in our setting over a horizon based on the average length of retirement.

as a result of the judge's decision being made with probability $\frac{\xi_d^M}{\Sigma_{i=d}^3 \xi_i^M}$.

Beyond differences in employment and assets, workers are distinguished by their skill-type *k* which determines the wage offer distribution and probabilities of distress, exogenous separation, and offer arrival (conditional on search effort) that the worker faces. Workers also face differences in local labor market conditions *M* which affect the search process and distress processes.

The worker's state is defined by the combination of (a, h, w, d, k, M, t, i) when employed and $(a, h, \tilde{w}, \tilde{h}, k, M, t, i)$ when unemployed, where *a* denotes liquid assets, *h* denotes current human capital, *w* denotes wage, *d* denotes distress status for employed workers, *k* denotes skill type, *M* denotes labor market conditions, and *t* and *i* denote the age and employment status of the worker. For unemployed workers, \tilde{w} and \tilde{h} denote the wage and human capital during the most recent period of employment, which determine the worker's UI benefit income. After observing her state, the worker chooses an optimal level of assets to carry into the next period, a^* , a level of consumption c^* , and a level of search effort e^* —all of which potentially depend on the full tuple of state variables—in order to maximize expected utility. To ease notation, we suppress the dependence of the policy functions a^* , c^* , and e^* throughout the remainder of this section.

At age t = T = 504—the last period before retirement—the worker's optimal search effort is trivially $e^* = 0$, because search is costly and the worker cannot accept offers to work in retirement in our model. The retiring worker's continuation value is given by the residual utility function $v(a_{T+1})$ which we use as a reduced-form way of capturing various motivations to save for retirement. The problem of a retiring employed worker is given by

$$V_T^1(a, h, w, d, k, M) = \max_{a^*, c^*} u(c^*) + \beta v(a^*)$$

s.t. $c^* + \tilde{q}a^* \le \omega_1 (w \cdot exp(h))^{\omega_2} + a$
 $c^* \ge 0$
 $a^* \ge \underline{a}$

and the problem of a retiring unemployed worker is given by

$$V_T^0(a, h, \tilde{w}, \tilde{h}, k, M) = \max_{a^*, c^*} u(c^*) + \beta v(a^*)$$

s.t. $c^* + \tilde{q}a^* \le \omega_1 (b(\tilde{w}, \tilde{h}))^{\omega_2} + a$
 $c^* \ge 0$
 $a^* > a$

where we let $\tilde{q} = q_s$ when $a^* \ge 0$ and $\tilde{q} = q_b$ when $a^* < 0$.

At ages t < T, unemployed workers have the ability to search for a job and accept any wage offer that yields higher value than their current state of unemployment. We assume that newly hired workers do not face any immediate risk of their firm becoming distressed. Thus the problem

of an unemployed non-retiring worker who searches for new employment is given by

$$\begin{split} V_{t}^{0}\left(a,h,\tilde{w},\tilde{h},k,M\right) &= \max_{a^{*},c^{*},e^{*}} u\left(c^{*}\right) - \psi\left(e^{*}\right) + \\ &+ \beta\lambda_{0,0,M}e^{*}\left(\mathbb{E}_{h',w',M'}\left[\max\left\{V_{t+1}^{0}\left(a^{*},h',\tilde{w},\tilde{h},k,M'\right),V_{t+1}^{1}\left(a^{*},h',w',0,k,M'\right)\right\}\right]\right) + \\ &+ \beta\left(1-\lambda_{0,0,M}e^{*}\right)\mathbb{E}_{h',M'}\left[V_{t+1}^{0}\left(a^{*},h',\tilde{w},\tilde{h},k,M'\right)\right] \\ s.t. \ c^{*} + \tilde{q}a^{*} &\leq \omega_{1}\left(b\left(\tilde{w},\tilde{h}\right)\right)^{\omega_{2}} + a \\ c^{*} &\geq 0 \\ e^{*} &\in [0,1] \end{split}$$

Unemployed workers who choose not to search for new employment face a nearly-identical problem, with the only difference arising in the worker's budget constraint. Receiving pre-tax UI benefit b(w,h) requires workers to be actively searching for a job, consistent with the policy environment that we model. In contrast, unemployed workers who do not actively search for new employment receive a basic income from a separate social assistance program—equal to \underline{b} post-tax—which does not depend on their prior earnings.

The problem of a non-retiring employed worker can be written similarly, though employed workers must also account for the probability $\pi_{k,M}$ that their firm becomes distressed and the probability $\delta_{k,M}$ of being exogenously separated from their employer. Thus the problem of an employed worker in a non-distressed (d = 0) firm can be written as

$$\begin{split} V_t^1\left(a,h,w,0,k,M\right) &= \max_{a^*,c^*,c^*} u\left(c^*\right) - \psi\left(e^*\right) + \\ &+ \beta \lambda_{1,0,M} e^* \left(\mathbb{E}_{h',w',d',M'}\left[\max\left\{V_{t+1}^1\left(a^*,h',w,d',k,M'\right),V_{t+1}^1\left(a^*,h',w',0,k,M'\right)\right\}\right]\right) + \\ &+ \beta\left(1 - \lambda_{1,0,M} e^*\right)\left(1 - \delta_{k,M}\right)\pi_{k,M} \mathbb{E}_{h',M'}\left[V_{t+1}^1\left(a^*,h',w,1,k,M'\right)\right] + \\ &+ \beta\left(1 - \lambda_{1,0,M} e^*\right)\left(1 - \delta_{k,M}\right)\left(1 - \pi_{k,M}\right)\mathbb{E}_{h',M'}\left[V_{t+1}^1\left(a^*,h',w,0,k,M'\right)\right] + \\ &+ \beta\left(1 - \lambda_{1,0,M} e^*\right)\delta_{k,M} \mathbb{E}_{h',M'}\left[V_{t+1}^0\left(a^*,h',w,h,k,M'\right)\right] \\ s.t. \ c^* + \tilde{q}a^* \leq \omega_1 \left(w \cdot exp(h)\right)^{\omega_2} + a \\ c^* \geq 0 \\ e^* \in [0,1] \end{split}$$

Workers in distressed firms form conditional beliefs about the likelihood that the decision will be made at the end of the current period. Workers in distressed firms may leave their current employer through on-the-job search or through the standard exogenous separation shock $\delta_{k,M}$. If the worker remains at the distressed firm at the end of the period and the judge's decision is made, then with probability \overline{L} the judge liquidates the firm and the worker enters the next period unemployed. With the complementary probability $1 - \overline{L}$, the case is dismissed and the worker remains at the current employer in the subsequent period. If the decision is not made at the end of the current period, then the worker remains in the distressed state, but revises her beliefs based on an additional period of being aware of distress. Therefore the problem of a worker in a distressed firm (d > 0) can be written as

$$\begin{split} V_{t}^{1}\left(a,h,w,d,k,M\right) &= \max_{a^{*},c^{*},e^{*}} u\left(c^{*}\right) - \psi\left(e^{*}\right) + \\ &+ \beta\lambda_{1,d,M}e^{*}\left(\mathbb{E}_{h',w',M'}\left[\max\left\{\tilde{V}\left(a^{*},h',w,k,M'\right),V_{t+1}^{1}\left(a^{*},h',w',0,k,M'\right)\right\}\right]\right) + \\ &+ \beta\left(1-\lambda_{1,d,M}e^{*}\right)\left(1-\delta_{k,M}\right)\left(1-\xi_{d}^{M}/\Sigma_{n=d}^{3}\xi_{n}^{M}\right)\mathbb{E}_{h',M'}\left[V_{t+1}^{1}\left(a^{*},h',w,d+1,k,M'\right)\right] + \\ &+ \beta\left(1-\lambda_{1,d,M}e^{*}\right)\left(1-\delta_{k,M}\right)\left(\xi_{d}^{M}/\Sigma_{n=d}^{3}\xi_{n}^{M}\right)\left(1-\tilde{L}\right)\mathbb{E}_{h',M'}\left[V_{t+1}^{1}\left(a^{*},h,w,0,k,M'\right)\right] + \\ &+ \beta\left(1-\lambda_{1,d,M}e^{*}\right)\left(1-\delta_{k,M}\right)\left(\xi_{d}^{M}/\Sigma_{n=d}^{3}\xi_{n}^{M}\right)\left(\tilde{L}\right)\mathbb{E}_{M'}\left[V_{t+1}^{0}\left(a^{*},h,w,h,k,M'\right)\right] + \\ &+ \beta\left(1-\lambda_{1,d,M}e^{*}\right)\delta_{k,M}\mathbb{E}_{M'}\left[V_{t+1}^{0}\left(a^{*},h,w,h,k,M'\right)\right] \\ s.t. \ c^{*}+\tilde{q}a^{*} \leq \omega_{1}\left(w \cdot exp(h)\right)^{\omega_{2}} + a \\ c^{*} \geq 0 \\ e^{*} \in [0,1] \end{split}$$

where $\tilde{V}(a^*, h', w, k, M')$ denotes the worker's value accounting for the possibilities of no decision being made (leading to distress status d' = d + 1), the firm being liquidated, the bankruptcy case being dismissed, and the worker exogenously separating from the firm for non-distress reasons.

In each of the preceding equations, expectations with respect to the offered wage w' are computed according to the offer distribution G(w'|k, M), expectations with respect to M' are computed according to the Markov process defined by the state-persistence probabilities τ_0 and τ_1 , and expectations with respect to h' are computed according to the transition matrix implied by Δ_h^+ and Δ_h^- given the worker's employment status. Where not explicitly written out, expectations with respect to d' imply that d' = 1 with probability $\pi_{k,M}$ and d' = 0 with the complementary probability $1 - \pi_{k,M}$.

D.2 Belief Updating Example

Workers form Bayesian beliefs about the probability of the bankruptcy judge's decision being made at the end of the current period. We assume that workers know the distribution $(\xi_1^M, \xi_2^M, \xi_3^M)$ where ξ_n^M denotes the share of workers who receive exactly *n* periods of advance notice of the judge's decision in market condition *M*. When workers first become aware that their firm is distressed, their prior belief about the probability that the judge's decision will occur in the current period is given by ξ_1^M . Workers update their beliefs each period that they remain distressed because the lack of decision in the preceding period is informative about what part of the distribution the worker falls in.²

As an example, suppose that the distribution of advance information about the firm's distress is given by $(\xi_1^M, \xi_2^M, \xi_3^M) = (0.5, 0.3, 0.2)$. Suppose that a bankruptcy petition is filed against the firm in period \tilde{t} and thus the bankruptcy judge decides the case at the end of period \tilde{t} , potentially

²This style of belief updating is similar to the belief updating behavior of agents solving an unobservable queueing problem under a first-come-first-serve service rule (see, e.g., Che & Tercieux (2021)).

leading to workers being laid off. Because we only allow workers to have up to three periods of advance notice of the pending judge's decision, no workers directly consider the threat of potential layoff due to liquidation prior to period $\tilde{t} - 2.^3$

In period $\tilde{t} - 2$, three periods before the judge's decision, a fraction $\xi_3^M = 0.2$ of workers become aware that the firm is in distress. These workers know that the only way that the judge's decision will be made at the end of the current period is if they only received one period of advance information. Thus, these workers believe that the judge's decision will occur in the current period with probability $\xi_1^M = 0.5$.

In period $\tilde{t} - 1$, two periods before the judge's decision, an additional share $\xi_2^M = 0.3$ of workers become aware that the firm is in distress. By the same rationale as in the previous period, these workers believe that the judge's decision will occur in the current period with probability $\xi_1^M = 0.5$. At the same time, workers who were aware of the firm's distress in period $\tilde{t} - 2$ but have not yet experienced a decision will update their beliefs. Because a decision did not occur in the first period that they were aware of the distress, these workers know that they cannot be part of the fraction $\xi_1^M = 0.5$ of workers who received one period of advance information. Moreover, they know that the judge's decision will only occur in the current period if they are part of the fraction $\xi_2^M = 0.3$ of workers who received two periods of advance information. Thus, they believe the judge's decision will occur at the end of the current period with the conditional probability $\frac{\xi_2^M}{1-\xi_1^M} = 0.6$.

In period \tilde{t} , one period before the judge's decision, the remaining share $\xi_1^{\tilde{M}} = 0.5$ of workers become aware that the firm is in distress, believing that the decision will occur in the current period with probability $\xi_1^M = 0.5$. At the same time, workers who first became aware of the firm's distress in period $\tilde{t} - 1$ update their beliefs as previously described, believing that the decision will occur in the current period with probability $\frac{\xi_2^M}{1-\xi_1^M} = 0.6$. Finally, workers who first became aware of the firm's distress in period $\tilde{t} - 2$ and who have not yet experienced the decision again update their beliefs, now knowing that they are not part of the fraction ξ_1^M of workers who receive one period of advance notice nor the fraction ξ_2^M of workers who receive two periods of advance notice. Thus, because we only allow up to three periods of advance information, these workers know with certainty that the judge's decision will be made in the current period, because $\frac{\xi_3^M}{1-\xi_1^M-\xi_2^M} = 1$.

Figure D.1 illustrates how the distribution of beliefs among workers in a firm changes as the firm approaches the period of the bankruptcy judge's decision. In particular, only fraction ξ_3^M of workers fully internalize the risk of the judge's decision at the time it occurs. The remaining workers have non-degenerate Bayesian beliefs based on their knowledge of the distribution $(\xi_1^M, \xi_2^M, \xi_3^M)$ and updated based on the number of periods that they have experienced distress without the judge's decision being made.

³Because we assume that workers have perfect foresight, workers prior to period $\tilde{t} - 2$ will still incorporate the possibility of becoming aware that their firm is in distress—leading to an eventual layoff possibility—into their expectations.

Figure D.1: Belief Distribution in Distress Cases



Notes: This figure illustrates an example of how the distribution of workers' beliefs about the probability of the judge's decision being made at the end of the current period changes as the firm approaches the actual time of the judge's decision. Workers who are notified more than one period prior to the judge's decision update their beliefs over time. By the time the judge's decision is actually made, all workers believe that the decision will occur with positive probability; however, only workers with the most advance information fully internalize the risk and believe that the decision will occur with certainty. Prior to being notified of the firm's distress, workers believe there is zero probability of facing a layoff due to the employer being shutdown in the current period.

D.3 Model Moments and Fit

In this section we describe the full set of moments that we target when calibrating structural parameters internally and present the calibrated values of the structural parameters along with the simulated moment values.

D.3.1 Targeted Moments

We target a set of 12 moments to internally calibrate 11 structural parameters of the model.⁴ Section 6.2 describes the main identification argument for each parameter. We report the target and simulated values of each moment in Appendix Table D.2.

Retirement Savings Because we do not model an explicit retirement period, we rely on a residual utility function $v(a_{T+1})$ to capture the value of entering retirement with a given level of assets. We include this term in the model primarily to ensure that workers do not exhaust their savings as they approach the end of the model's finite horizon, as this would distort the model's ability to capture the role of private assets as an insurance mechanism. We target the ratio of workers' net wealth at age 67 to workers' net wealth at age 64, which summarizes the relative level of savings that occurs near the end of the life-cycle and provides identifying variation for the scaling coefficient φ in the residual utility term.

We compute the target value using our overall population of workers by computing the average net wealth of all workers at age 67 and the average net wealth of all workers at age 64 and taking the corresponding ratio. Because the model has only liquid assets, we exclude the effects of passive savings through the ownership of housing (i.e., the change in net wealth attributable to the change in housing prices) in order to avoid forcing the model to fit excessive levels of saving in liquid assets.

⁴We explicitly calibrate 11 of the 13 unassigned structural parameters. The values of the remaining 2 parameters are implied by the constraints that ξ_1^M , ξ_2^M , and ξ_3^M must sum to 1 for both values of *M*.

Aggregate Employment Transition Rates In addition to differences in the choice of search effort, workers in the model face different offer arrival rates conditional on their employment status. In order to ensure that these offer rates fit with what we observe empirically, we include the aggregate transition rates from unemployment to employment (UE) and from employment to employment (EE) as targeted moments in the model. While we primarily use these moments to identify the maximum offer arrival rates λ_i , these moments also have a significant effect on workers' choice of effort and thus exert some influence over our choice of the scale of search costs.

We measure transition rates among our overall sample of workers, excluding those who are classified as not participating in the labor force (which affects our estimate of the UE rate). We calculate analogous values in the model, where we conclude that workers are not participating in the labor force if they are unemployed and choose effort $e^* = 0$. We calculate the aggregate UE and EE rates in the model by taking a weighted average of the UE and EE rates in strong and weak market conditions separately, because the number of periods we simulate in each market condition inside the model may not be exactly proportional to the data.

Precautionary Search Because we are interested in using our structural model to asses the role of on-the-job search as an insurance mechanism, we want to ensure that the precautionary search behavior of workers in the model matches what we observe empirically. Workers' ability to engage in precautionary search is significantly influenced by the availability of advance information about the pending layoff, which lowers the value of the current employment position and makes search relatively more attractive. To determine the potential duration of advance information, we analyze how EE rates vary in the months leading up to and immediately following the bankruptcy petition and subsequent judge's decision and interpret excess EE separations as evidence that workers are aware of the firm's distress.

We target the share of workers who leave the distressed firm through an EE transition in each aggregate state, with the understanding that the majority of such "precautionary" transitions out of the distressed firm occur at the time the bankruptcy petition is filed or just before due to advance information. These moments provide identifying variation that helps determine the distribution of advance information defined by ξ_i^M .

Effects of Liquidation on Labor Market Outcomes We target several of our 2SLS estimates from Section 4.We include the average effect of liquidation on employment in year 4 to capture the fact that the effect does not fully diminish within the four year horizon. To provide further structure to how liquidation affects workers' employment outcomes, we measure the effects of liquidation on both wage income and UI benefit income. We target the relative change in wages in year 1 and year 4, which we measure as the ratio of the estimated effect of liquidation on wage income to the mean wage income of untreated compliers. Similarly, we target the relative change in UI benefit receipts in year 1 and year 4, which reinforces the effect of employer shutdown on employment and which depends in part on which workers are subject to unemployment spells and which workers

instead make employment-to-employment transitions before the judge's decision is made. We also target the average effect of liquidation on wage income from year 0 to year 4 separately by local labor market conditions, which provides further evidence which workers are able to engage in precautionary employment transitions and which workers are subject to layoff as a result of the firm being shut down.

D.3.2 Calibration Results

Table D.1 describes the model parameters which we set externally, either based on estimates from the literature or by estimating values directly from our data. The source of each value is discussed in Section 6.2.

Table D.2 presents the result of our joint calibration of the remaining model parameters as well as the values of the 12 moments that we target. For each moment, we report the simulated and target values, as well as the weight we give to the moment when minimizing the sum of squared residuals in our Method of Simulated Moments estimator. We weight moments replicating our 2SLS estimations of the effect of layoff according to their t-statistics. Population moments describing the overall sample of workers are given weight of one. The rate of precautionary transitions out of distressed firms in each state is given a weight of ten in order to ensure that the model replicates the use of search as a form of self insurance.

In sum, we find that the model does a good job of fitting the insurance behavior of workers compared to the targets. In particular, we generate relatively accurate values of the share of workers who leave distressed firms through EE transitions prior to the judge's decision in each state. In addition, the effects of liquidation on workers' labor market outcomes fall within the 95% confidence intervals of our empirical estimates for all targeted moments.

Panel (a): Calibrated Parameters	Parameter	Value
Coefficient of Relative Risk Aversion	γ	1.700
Subjective Discount Factor	β	0.9975
Bond Price (Savings)	q_s	1.000
Bond Price (Borrowing)	q_b	0.993
Borrowing Limit at Retirement	<u>a</u>	-593.617
Distress Risk (Low Skill, Strong Market)	$\pi_{0,0}$	0.00080
Distress Risk (High Skill, Strong Market)	$\pi_{1,0}$	0.00035
Distress Risk (Low Skill, Weak Market)	$\pi_{0,1}$	0.00103
Distress Risk (High Skill, Weak Market)	$\pi_{1,1}$	0.00018
Separation Risk (Low Skill, Strong Labor Market)	$\delta_{0,0}$	0.0208
Separation Risk (High Skill, Strong Labor Market)	$\delta_{1,0}$	0.0089
Separation Risk (Low Skill, Weak Labor Market)	$\delta_{0,1}$	0.0193
Separation Risk (High Skill, Weak Labor Market)	$\delta_{1,1}$	0.0088
Probability of Liquidation	Ē	0.553
Scale of Taxation	ω_1	5.366
Progressivity of Taxation	ω_2	0.844
UI Replacement Rate	r	0.624
Maximum UI Benefit (monthly NOK1,000)	$ar{b}$	28,080
Minimum Social Assistance Income (monthly NOK1,000)	<u>b</u>	8,333
Scale of Wage Distribution (Low Skill, Strong Market)	$\mu_{G}^{0,0} \\ \mu_{G}^{1,0} \\ \mu_{G}^{0,1} \\ \mu_{G}^{0,-} \\ \sigma_{G}^{0,-} \\ \sigma_{G}^{1,0} \\ \sigma_{G}^{0,1} \\ $	13.054
Scale of Wage Distribution (High Skill, Strong Market)	$\mu_{G}^{1,0}$	13.007
Scale of Wage Distribution (Low Skill, Weak Market)	$\mu_G^{0,1}$	13.038
Scale of Wage Distribution (High Skill, Weak Market)	$\mu_G^{1,1}$	12.961
Shape of Wage Distribution (Low Skill, Strong Market)	$\sigma_{C}^{0,-}$	0.176
Shape of Wage Distribution (High Skill, Strong Market)	$\sigma_C^{1,0}$	0.206
Shape of Wage Distribution (Low Skill, Weak Market)	$\sigma_{C}^{0,1}$	0.183
Shape of Wage Distribution (High Skill, Weak Market)	$\sigma_G^{I,1}$	0.250
Labor Market Tightness (Non-Distress, Strong Market)	$\theta_{0,0}$	1.000
Labor Market Tightness (Distress, Strong Market)	$\theta_{1,0}$	0.578
Labor Market Tightness (Non-Distress, Weak Market)	$\theta_{0,1}$	0.234
Labor Market Tightness (Distress, Weak Market)	$\theta_{1,1}$	0.178
Probability of Remaining a Strong Market	$ au_0$	0.999
Probability of Remaining a Weak Market	$ au_1$	0.961

Table D.1: External Model Parameters

Notes: This table shows the values of structural model parameters which we take externally. Preference parameters such as the coefficient of relative risk aversion and subjective discount factor are taken from common estimates in the literature. Institutional details such as the UI replacement rate are based on the empirical institutions. The remaining parameter values are estimated directly from the data, either using the overall sample of workers or the sub-sample of workers who are employed by firms undergoing bankruptcy proceedings. Labor market tightness values are combined with the internally calibrated maximum offer rates λ_i to produce the full set of offer arrival rates $\lambda_{i,d,M}$.

Panel (a): Calibrated Parameters	Parameter	Value	
Scale of Retirement Savings Motive	φ	49.660	
Scale of Search Cost (Unemployed)	μ	2.930	
Elasticity of Search Cost	η	7.630	
Distribution of Advance Info (Strong Market)	$(\xi_1^0,\xi_2^0,\xi_3^0)$	(0.219; 0.045; 0.736)	
Distribution of Advance Info (Weak Market)	$\left(\xi_1^{\overline{1}},\xi_2^{\overline{1}},\xi_3^{\overline{1}}\right)$	(0.523; 0.220; 0.257)	
Offer Arrival Rate (Unemployed)	λ_0	0.313	
Offer Arrival Rate (Employed)	λ_1	0.900	
Pro. of Gaining Human Capital (Employed)	$\Delta^h_+ \Delta^h$	0.067	
Prob. of Losing Human Capital (Unemployed)	$\Delta^{\dot{h}}_{-}$	0.815	
Panel (b): Targeted Moments	Simulation	Data	Weight
Aggregate UE Rate	0.091	0.173	1.000
Aggregate EE Rate	0.035	0.012	1.000
Age 67 to Age 64 Savings Ratio	1.656	1.137	1.000
Effect of Layoff on Any Employment (Yr 4)	0.004	-0.099	1.269
Relative Effect on Wage (Yr 1)	-0.382	-0.277	2.079
Relative Effect on Wage (Yr 4)	-0.082	-0.158	1.378
Relative Effect on UI Income (Yr 1)	2.257	2.229	3.206
Relative Effect on UI Income (Yr 4)	-0.029	1.023	0.706
Relative Effect on Wage (Year 0-4; Strong Market)	-0.174	-0.091	0.444
Relative Effect on Wage (Year 0-4; Weak Market)	-0.243	-0.211	2.111
Precautionary Transition Rate in Strong Market	0.376	0.346	10.000
Precautionary Transition Rate in WeakMarket	0.167	0.137	10.000

Table D.2: Model Fit

Notes: This table shows the results of our joint calibration of structural model parameters. Panel (a) reports the calibrated value for each structural parameter that is part of the joint estimation. Panel (b) reports the simulated value of each of our targeted moments, the associated target value, and the weight we assign to the moment in our evaluation of the estimator. We choose the vector of structural parameters in Panel (a) to minimize the weighted sum of squared residuals based on the simulated moment values reported in Panel (b).

D.4 Decomposing Wage Losses

To understand the mechanisms, we assess the contribution of human capital, the job ladder, search frictions, and the disutility cost of search to wage losses. The most closely related frameworks are Jarosch (2022), who incorporates heterogeneous layoff risk in a model with on-the-job search, and Lise (2013), who incorporates risk aversion into a job ladder model. Like existing frameworks, workers recoup some lost earnings as they re-climb the ladder and rebuild their human capital.

The key innovation of our framework is that upon learning about heightened layoff risk, workers start accepting outside offers with a lower wage, as the increased layoff risk decreases the reservation wage before the separation occurs, generating the endogenous selection of workers in distressed firms.

Figure D.2 provides a graphical decomposition of the simulated effect of liquidation on wage income. We begin by constructing a counterfactual where workers retain their previous level of human capital when they find new employment. This effect is illustrated by the dark grey area, leading to persistent wage income loss that does not fully diminish by the end of the four-year horizon. Our next counterfactual additionally assigns the worker the same position on the wage ladder upon re-employment. We see that position on the wage ladder has a slightly larger relative effect at the end of the period, as illustrated by the light shade.



Figure D.2: Decomposed Effect on Wage Income

Notes: This figure shows how search frictions, the costs of search effort, the wage ladder, and human capital contribute to the effect of liquidation on wage income in year 0 through year 4 in our fitted baseline model. We construct a series of counterfactual estimates of the effect of liquidation on wage income where we progressively remove these mechanisms from the model by allowing workers to recover their prior level of human-capital and pre-tax wage upon becoming re-employed after liquidation and by exogenously setting the probability of receiving an offer to the maximum probability $\lambda_{i,d,M}$ or by guaranteeing an offer. The reduction in the estimated wage effect as each component is progressively removed determines the effect of that mechanism on our baseline estimate.

Figure D.3 demonstrates that average human capital levels are not monotonically increasing in employer quality, suggesting that workers do not traverse the whole wage ladder "rung by rung"; rather, workers employed by the highest quality firms are hired directly from relatively low-quality firms or unemployment, while workers who move up the ladder "rung by rung" find it optimal to stop searching before reaching the highest quality employers. This dynamic makes workers in the highest quality firms less likely to naturally recover their pre-liquidation positions, exacerbating the wage effect.

Next, we decompose the wage effect by separating the two components contributing to the employment-driven effect on wages. We estimate an additional counterfactual effect of liquidation on wage, assuming that the probability of receiving an offer is exogenously given by the maximum

offer rate $\lambda_{i,d,M}$ that we calibrated in the model and assuming that any offer a worker receives yields the same pre-tax wage and human capital as the worker's terminated employer. Finally, we estimate the wage effect when the worker is immediately re-offered their pre-liquidation position (with the same pre-tax wage and human capital) after the liquidation occurs. When both search frictions and costly effort are present, the employment-driven wage effect diminishes by year 3. When only search frictions are present, the effect diminishes slightly more rapidly.



Figure D.3: Skill Premia by Employer Quality

Notes: This figure plots the average skill premium from human capital by employer quality (base wage) for workers who leave the distressed firm prior to the judge's decision and for workers who remain at the distressed firm at the time of the judge's decision.